Contents lists available at ScienceDirect



Solar Energy Materials and Solar Cells

journal homepage: www.elsevier.com/locate/solmat

Natural energy materials and storage systems for solar dryers: State of the art

Bade Venkata Suresh^a, Yegireddi Shireesha^a, Teegala Srinivasa Kishore^b, Gaurav Dwivedi^c, Ali Torabi Haghighi^d, Epari Ritesh Patro^{d,*}

^a Mechanical Engineering, GMR Institute of Technology, Rajam, 532127, India

^b Electrical & Electronics Engineering, GMR Institute of Technology, Rajam, 532127, India

^c Department of Energy, Maulana Azad National Institute of Technology, Bhopal, 462003, India

^d Water, Energy, and Environmental Engineering Research Unit, University of Oulu, FIN-90014, Oulu, Finland

ARTICLE INFO

Keywords: Solar dryers Thermal energy storage devices Agro produce Fruit produce Natural energy materials

ABSTRACT

Inappropriate food conservation practices along with insufficient and inefficient storage systems are often responsible for the deterioration in food quality of agricultural products leading to food insecurity and economic loss. Open sun drying is a well-known traditional food preservation technique but is limited in use due to its low efficiency and long drying times. Solar drying is considered the most effective, economical, green and sustainable technology available to preserve farm produce. In this regard, an attempt has been made in this study, to review the solar dryer technologies, natural energy materials and storage systems available for persevering food products and reported in detail. An extensive classification and comparative analysis of solar dryers have been presented. Evolutionary classification and performance assessment using various indicators has been carried out for solar dryers employing natural energy materials for energy storage. Polices, challenges, risks and recommendations for the improvement of existing solar dryers as well as the development of new technologies impacting the quality, economic, social and environmental aspects of solar dried products have been discussed in detail. The study concluded that solar drying processes with thermal energy storage devices based on natural materials are most preferred for delivering extended shelf life for farm produce in an energy-efficient and sustainable manner.

1. Introduction

To ensure food security for the ever-growing global population, food wastage must be monitored and reduced during harvesting, handling, promoting, and circulation. Inadequate storage units and handling processes cause the flavor, color, taste and quality of food items to deteriorate. Many developing nations are losing a lot of agricultural food production and allied products owing to the above reasons. The overall post-harvest losses in vegetable and fruit produce are estimated to be 30% to 40% of the total production which is one of the prime reasons for the rise in inflation in agriculture products [1,2]. In the United States, 30% of food, totalling US\$48.3 billion (\in 32.5 billion), is wasted annually. Since water is the main source of farming, it is estimated that almost half of the water utilized to create this food is also lost. Depending on the sector, farm-level losses are probably between 15 and

35%. Unexpectedly super-markets, trailing about 1%, and the retail sector have rather significant loss rates of roughly 26%. Losses total between US\$90 billion and US\$100 billion annually [3]. 300 million people might be fed with the food that is presently lost or wasted in Latin America [4]. An estimated 6.7 million tonnes of food are wasted by British homes each year, or around one-third of the 21.7 million tonnes that are bought. This indicates that over 32% of the food purchased each year is wasted. Currently, municipal governments collect the majority of this (5.9 million tonnes, or 88%). If it had been handled more effectively, the majority of the food waste (4.1 million tonnes, or 61%) might have been consumed [5,6]. Europe's current food waste can feed 200 million people [4].

The Australia Institute surveyed more than 1600 Australian families and found that \$10.5 billion was spent on things that were either never utilized or were thrown away nationwide. This comes to more than

* Corresponding author.

https://doi.org/10.1016/j.solmat.2023.112276

Received 2 December 2022; Received in revised form 22 February 2023; Accepted 26 February 2023 Available online 21 March 2023

0927-0248/© 2023 The Author(s). Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).



E-mail addresses: suresh.bv@gmrit.edu.in (B.V. Suresh), y.shireesha@gmrit.edu.in (Y. Shireesha), kishore.ts@gmrit.edu.in (T.S. Kishore), gdiitr2005@gmail.com (G. Dwivedi), ali.torabihaghighi@oulu.fi (A.T. Haghighi), ritesh.patro@oulu.fi (E.R. Patro).

\$5000 per person every year [7]. Food cereal post-harvest losses in several African nations are reported to be 25% of the whole produced crop. Post-harvest losses for some crops, such as fruits, vegetables, and root crops, which are less resilient than cereals, can exceed 50% [8]. Economic losses in the dairy industry from spoilage and waste could amount to US\$90 million on average per year in East Africa [9]. Every year, Kenya loses over 95 million litres of milk, which is worth about \$22.4 million. In Tanzania, annual losses total around 59.5 million litres of milk or more than 16% of total dairy production during the dry season and 25% during the rainy season. About 27% of the milk produced in Uganda is wasted, costing the country US\$23 million annually [9]. Africa's current food loss could feed 300 million people [4]. The Food Corporation of India (FCI) reports that losses for cereals and oil seeds are smaller, at 10-12%. A total estimated worth of 240 billion Rupees is wasted annually in the form of food cereals (23 million tons), fruits (12 million tons) and vegetables (21 million tons). According to a recent estimate by the Ministry of Food Processing Industries, India, the country trashes agro-products valued at 580 billion Rupees annually **[10]**.

Globally, each year, almost 1.3 billion tonnes, or 1/3 of food cultivated for human feeding worldwide, is squandered. Food waste and losses total around US\$ 310 billion in underdeveloped nations and US\$ 680 billion in wealthy nations. The amount of food wasted by both developing and industrialized nations is approximately equal to 670 million tons. Vegetables, tubers, fruits and roots are among the major agricultural products having the maximum depletion rates of food. The global food-waste rate is about 30% for cereals, 45% for root crops, 20% for fruits, 20% for vegetables, 20% for oilseeds, 20% for meat, 20% for dairy products and 30% for fish. Consumer waste per person varies between 95 and 115 kg annually in North America and Europe, compared to 6–11 kg in Southeast & South Asia and Sub-Saharan Africa. Overall food cultivation for human feeding per capita is approximately 900 kg/year in developed countries, which is nearly twice the food production in underdeveloped countries (460 kg/year). In industrialized countries, more than 40% of losses occurring at the retail and consumer levels, compared to 40% of losses in poor countries at the post-harvest and processing stages.

Food loss and wastage lead to a substantial waste of natural resources, which includes land, labour, water, capital and energy. Additionally, because of this waste unnecessary emissions like greenhouse gas levels are increasing in the environment, which leads to climate change and global warming. Large amounts of food are wasted at the retail level as a result of quality standards that place an excessive emphasis on appearance. If 1/4 of food that is presently wasted or lost worldwide might be salvaged, 870 million hungry people would be able to eat. Food waste in poor nations primarily occurs at the beginning and is caused by technical, managerial, and financial limitations in harvesting methods as well as cooling and storage facilities. Famine is a big concern, according to the Food and Agriculture Organization (FAO), population expansion and management of waste food (\cong 1.3 billion tonnes) can enhance the critical consequences. This is almost 30% of world nutrition manufacturing, which is thrown away every year [11, 12]. Food loss and waste could be decreased by bolstering the supply chain through direct support for farmers and investments in transportation, infrastructure, and an expansion of the food and packaging industries. Food is lost and wasted primarily after the supply chain in medium and high-income nations. In contrast to the situation in developing nations, consumer behavior has a significant impact on developed nations. Literature research revealed that lack of cooperation amongst supply chain entities is a contributing element. Agreements between farmers and buyers may help to improve coordination. The quantity of losses and waste can be reduced by increasing attentiveness between consumers, retailers and businesses as well as by defining the use of food which is currently wasted [13,14].

Drying food products is the most sought-after method for preserving agricultural produce. The conventional drying technologies presently

used in the food industry are hot-air-based systems that are normally considered energy intensive processes with higher Greenhouse Gas (GHG) emissions. The popular conventional drying methods are hybrid drying systems, heat pump drying, super-heated steam drying, vacuum drying, microwave drying, refractance window dehydration method, electro technologies, ohmic heating/drying, adsorption mediated drying systems and impingement drying. Around 35-45% of the energy input is wasted as the hot exhaust gases, incur high energy losses and emit high GHG emissions [15,16]. The other drawbacks of conventional drying systems are longer drying times, non-uniform exposure to the temperature and hardening of the food materials. To avoid these issues, the food industry is motivated to work on the improvement of the existing drying technologies and the development of alternative drying technologies. It is expected that the new generation of dryers and drying technologies promote more sustainable development with higher thermal and energy efficiency, lower operation costs and improved product quality. With the world progressing towards the application of green technology and sustainable methods, the use of alternative energy-efficient drying systems is of paramount importance. Due to the above reasons, solar drying processes are found to be the best suitable methods for the dehydration of food products and eliminate the environmental issues rising due to conventional drying processes [17,18]. In this context, an extensive review of solar dryers and natural materials for thermal energy storage has been carried out in this paper and the subsequent sections are organized as follows. Section 2 deals with the classification and review of solar dryers along with their comparative analysis. Section 3 deals with various energy storage systems used in solar dryers. Section 4 analyses the performance of various natural materials used for energy storage in solar dryers along with evolutionary classification of these materials. The various performance indicators used in the performance assessment of solar dryers were discussed in Section 5. Section 6 discusses the quality, economic, social and environmental impact of solar dried products. Polices, challenges, risks and recommendations for the improvement of existing solar dryers as well as development of new technologies were discussed in Scetion7 and finally, Section 8 highlights the salient features discussed in this paper as conclusions. The study concludes that solar drying processes with thermal energy storage devices based on natural materials are most preferred for delivering extended shelf life for food production in an energy-efficient and sustainable manner.

2. Solar drying technologies

Drying is a process of food preservation that includes evaporating water from harvested products to prevent the growth of bacteria. Owing to the free and abundant energy from the sun, open solar drying is one of the oldest known food processing and storage strategies for drying agrogoods. However, technological advancement led to thermal processes for use in agro-industry for processing and storage of food [19]. The inherent drawbacks of conventional energy sources and the benefits associated with solar energy such as cost effectiveness and efficiency provided long-term solutions for ecological development. Solar radiation was abundantly available globally, which replaced the traditional way of drying for preserving fruits and vegetables for a maximum number of days. Food drying is also a difficult process that involves inconsistencies in heat and mass transmission, as well as physical and chemical reactions that might impair product quality [20,21]. It is a difficult process that includes simultaneous mass and heat transmission along with physicochemical changes. A crucial step in the production of many different kinds of chemical products is the adjustment and control of moisture levels in solid materials by drying. Since it is employed in almost every plant and facility that produces or processes solid materials, in the form of powders and granules, drying solid materials is one of the most popular and significant unit operations in the chemical process industries. The quality of the final product and the effectiveness of the drying operations can both be significantly impacted by the

chemical process industries [22,23].

Food drying is a diversified field as different foods require different types of drying processes. A natural or manufactured substance can be dried to remove surplus water and bring it up to the required moisture content. It requires a lot of energy to operate. Reducing the moisture level of foods is especially important because they often have substantially larger water contents than those appropriate for long-term storage. Enzymes, bacteria, yeasts, and moulds take longer to develop when there is less moisture in the meal. As a result, food can be kept and stored for a long period without going bad. The process of completely removing moisture from food until there is none left is another example of drying. When ready for usage, dehydrated food is rehydrated and nearly returns to its original state [24,25]. For agricultural products, especially those produced in medium to small quantities, solar energy drying is a fairly cost-effective method for preserving extra output. It is eco-friendly and it is still used to dry foodstuffs, agricultural products and crops on a domestic up to small commercial scale, greatly boosting the income of small agricultural farms and communities. Based on this, a variety of solar drying processes have evolved as follows [26]: a) Natural or Direct Sun Drying, b) Indirect sun drying and c) Mixed mode sun drying. Fig. 1 illustrates the classification of solar dryers into several categories.

Open sun drving (OSD) using Open drvers is the process of drving materials which are positioned on trays of thin layers, ground, concrete floors or mats. The process of exposing the drying materials to open wind and sun is illustrated in Fig. 2 [27]. It is the best traditional and favorable method because OSD is not dependent on other sources of energy, no additional provisions are required for drying products, is the most cost-effective process and skilful labor is also not required. However, OSD has several drawbacks, prompting researchers to hunt for alternatives. The potential for product damage from birds, insects, and animals is one of the main drawbacks. Rain, air humidity, and dew can all cause product degradation. Loss as a result of over-drying, spoilage due to the proliferation of microorganisms etc. can take place. A slightly modified method of OSD is natural or direct solar drying which is also considered traditional for dehydrating agricultural produce. It consists of a translucent cover, as illustrated in Fig. 3, which is intended to diminish temperature losses and to safeguard the foodstuff from rain and dust [28,29]. Its working principle is based on a direct solar dehydration strategy, and it is used to dry a variety of items, as shown in Table 1. A portion of the solar energy reflected on the transparent cover is transmitted within the box dryer, while the remainder is reflected.

Furthermore, the food surface reflects the amount of solar radiation. The temperature of the substance rises, and it dehydrates. The air blasted into the cabin drier removes the moisture in the food substance [30,31]. The indirect method of sun dehydration is more competent than the direct dehydration method, illustrated in Fig. 4, and it is used to dry a variety of items, as represented in Table 2. Through indirect sun dehydration, ambient air temperature is raised with a flat-plate collector in this method. Forced-convection and Natural-convection systems are the two types of indirect sun dryers, depending on how the air is passed over food. Material quality necessitates low temperatures. The disadvantages of direct solar dehydration are mitigated by using an indirect solar dehydration method [32-34]. Caramelisation and localized heat damage are rare since sun radiations are not openly striking the agro-products. These are ideal for preserving vitamins as well as color in highly pigmented goods. A flexible insulated conduct is commonly used to connect the separate air heater and drying rooms [35]. The north side of the drying chamber has a door that allows loading. The air collector can be angled to capture the optimum amount of solar energy at the time of year when the dried crop is collected.

Mixed mode drying generates the necessary heat for the drying process by combining the effects of solar radiation impacting the material to be dried with air heated by a solar collector. Solar dryers have used a range of phase transition and sensible thermal energy storage methods, with a focus on the usage of rock beds [36]. Many self-contained solar dryers do not require supplementary heating since thermal mass enables short-term storage of heat to allow drying processes even in bad weather [37]. Agro-goods which are quickly drying are the most suitable products for thermal energy storage methods, as thermal mass is hardly enough to withstand the overnight drying process [38,148]. Agricultural materials are dried in a hybrid dryer using direct sun radiation and backup heat stored in the event of a power outage. A variety of items are dried in this process, as shown in Table 3.

Alternative supporting energy sources, such as diesel, LPG, electricity, biomass and solar energy P–V module are used to pre-heat the air. It's possible to use this in the single dryer and mixed-type dryers like direct-type and indirect-type dryers [170]. Hybrid drying has the potential to reduce material microbe contamination [39,40,171]. In comparison to conventional electric resistant dryer, a heat pump drier can save up to 40% on energy [41]. In a study of banana drying, researchers discovered that hybrid drying takes less time than artificial or open drying [42,43]. And also, when related to the open dehydration process



Fig. 1. Classification of solar drying process.



Fig. 3. Natural or Direct solar drying Process.

(Solar-drying), hybrid dried bananas had improved color, aroma, and texture [44]. Another hybrid portable tunnel drier uses a solar non concentrating type collector (Flat palte) and solar system P–V module to improve the drying performance of the peppermint [45,172]. They discovered that hybrid-dried peppermint had a greater quality than open-air-sun-dried 'peppermint'. These hybrid dryers had diminished 32 tonnes of CO₂ and 31% productivity over their lifetime. The hybrid dryer has also been used on a variety of agro-goods, including cashew nuts, pineapples and mushrooms [46,47]. It is ideal for drying crops quickly while maintaining great product quality.

The majority of the studies looked at individual solar drying methods; some looked at indirect methods, while others looked at direct methods [48,49]. Meanwhile, other studies used experimental and numerical methods to evaluate the various forms of dehydration performance to determine their benefits and downsides, as well as to identify

future research possibilities. These studies, on the other hand, examined the various types of solar dryers in depth, focusing on their uses and benefits. In addition, this study examines solar-dryers with thermophysical features of several "Natural Energy Storage Materials". Solar dryers' performance in "direct, indirect, hybrid-mode and mixed-mode" is examined by considering and without considering natural-energy-storage materials, is examined. Also examined are the performance of commonly used natural energy storage materials such as water, pebbles, limestone, rocks, concrete, sand, quartz, gravel bricks, soil, clay and sandstone. Solar-air accumulating system dehydrating performances are highlighted as mechanical needs for improving solar dehydration technologies for food. The solar drying process has different advantages and as well as disadvantages, similarly different types of dries, and various dried products. The key observations for the review conducted in this section have been summarized and tabulated in

Solar Energy Materials and Solar Cells 255 (2023) 112276

Table	1					Table	1 (continued)				
Severa	Type of	gro-product Dried	s by applying vario Observations	us direct solar-dyer Remarks	rs. Ref	S. No.	Type of Solar Dryer	Dried Product	Observations	Remarks	Ref
<u>No.</u>	Solar Dryer Direct Solar Dryer	Banana	The models of Pabis and Henderson were shown to be the best fit for	The bananas took 21 h to dehydrate.	[84]				pineapple at a time.	pineapple and dried in a die solar tunnel drier, and the dried pineapple was of great	
			explaining banana sun dehydration			8	Solar Tunnel Dryer	Fish	Silver jewfish dried from 71 56% to	quality Solar tunnel dryer process is more efficient	[91]
2	Direct Solar- Dryer:	Сосоа	The grade analysis revealed that a 20 kg treatment was capable of producing high- quality food. Brown beans	This is the utmost basic low- capacity solar- dryer.	[85]				14.75% on 3 days in the solar tunnel drier, compared to 71.56%–23.63% in 3 days in open sun dehydrating.	than the open sun drying technology which is consuming less time to remove moisture from silver iewfish	
			were found in greater abundance in the lower loading regimens.			9	Solar Assisted Drying	Tea	Fresh leaves of tea are dried at a rate of $15.1 \text{ m}^3/$ min and it is observed that primary moist	Quick dehydration process for tea leaves is not acceptable due to the loss of key	[92]
3	Solar Assisted Heat Pump	Corn	In 240Hrs, lower the materials moisture content from almost 16%–14% (w.b). To reduce the moisture level by 1%, 1.24 kW were consumed for every grain tonne.	One of the best methods for dehydrating using a heat pump with low consumption of power which is 1.24KW/tonne.	[86]				concentration is reduced from 87% to 54%. Dehydration at a higher temperature causes the key element in tea to be lost, resulting in the loss of medicinal efforts	medicinal elements,	
4	Direct Solar Dryer	Mango	the sun for 12Hrs to reduce the moisture content from 95% to 13% (w.b).	efficient, and it takes longer to dry than open sun drying.	[8/]	10	Solar Tunnel Dryer	Apples	The open-sun dehydration reduced the moist concentration by	The solar tunnel dryer worked best for drying apples, whereas open-air drying	[93]
5	Natural Convection Solar Dryer	Onions	After 43Hrs and 32Hrs, the moisture content of full-size and cut onions drops	Chopped onions are dehydrating faster than full size onions.	[88]				32%–11% in 32Hrs, however, the solar tunnel dryer took just 28Hrs.	took a lengthy time.	
			from 35% to 57% and 6%, respectively. After 24Hrs, the moisture content of preserved onions reaches 46% for full size and 6% for chopped onions.			11	Solar Cabinet Dryer	Chilies	Solar dehydrating reduced the moisture content from 06% to 4% in only 9 days, but open-sun drying took 13- days to dehydrate 12 kg	When relating to open-sun drying and cabinet- drying, the time consumed by cabinet-drying is less than open- sun drying. Conventional solar dryers are	[94]
6	Solar Tunnel Dryer	Tomatoes	Tomato moist concentration almost diminished from 11 kg to 0.1 kg of	A combination solar dryer is recommended for tomatoes. Insects, rain, and	[89]				of chilies.	better at predicting temperature and moisture content at a steady pace.	
			water per kilogram when it is dried for 101 h in the open sun, but it takes only 86Hrs in the	dust were removed from the samples dried in the solar tunnel drier, and the dried samples		12	Natural Convection Solar Dryer	Chilies	Dehydrating 2 kg of chilies took 4 days in the solar dryer versus 15 days in the open-sun.	Concentrated drying process is more efficient than the conventional process.	[95]
_			solar tunnel dryer.	were of outstanding color and hygienic quality.	5002	Table dryers	4 [173]. Th like natural,	e advanta indirect a	ges and disadvan nd mixed mode c	tages of different lrying processes	nt sola are di
1	Solar Tunnel	Pineapple	This dryer can	Rain, insects, and	[90]	Clissec	by represent	ting their l	evel of constructi	on dehydration r	ate an

solar re discussed by representing their level of construction dehydration rate and economy.

handle 120

kg-150 kg of

dust were kept

out of the

Dryer



Fig. 4. Indirect solar drying.

3. Energy storage systems in solar dryers

The preservation of agro-products necessitates the use of a drying process. Hot-air at a range of temperatures (45 °C – 60 °C) is required for the safe dehydrating process, particularly for vegetables and fruits. Drving agro-products under controlled humidity and temperature settings allow them to dry quickly to the safest moisture level while maintaining optimum product quality. Solar-dryer system is coupled to a thermal backup division in hybrid solar thermal dryers. Thermal energy can be stored as a change in the material's internal energy, phase change material (PCM), thermochemical heat, or a mixture of these in well-insulated materials [50,174]. When there is no solar energy available, thermal energy dryers are used to keep the dehydration process going while maintaining a steady state temperature of the air. Heat energy storage forms a critical basis because it permits solar energy (also other renewable energy sources) with intermittent energy nature to balance the demand and supply. There are several methodologies for energy storing in numerous ways, including thermal, electrical, and mechanical [51–53]. Thermal energy can be stored as thermo-chemical heat, practical heat, latent heat a change and core energy of material in fine-shielded liquids or solids. Thermal storage is highly dependent on the usage of the type of solar-collector and demand. The solar air collector's estimated employed air temperature is crucial in determining the storage material's range [175]. To pick the precise material for heat storage, the thermo-physical parameters of the storage materials are regulated based on the temperature of the air [54-56]. The basic strategies for solar thermal energy storage are summarized in Fig. 5.

Agricultural and commercial goods can be continually dried using solar dryers based on thermal energy storage materials at steady state temperatures between 40° C - 60° C. Due to their use of clean energy resources and affordability, these dryers have gained popularity as a potentially feasible alternative to solar dryers that use fossil fuels [176]. These dryers use storage materials that can store energy during daylight hours and deliver that energy throughout the night. It plays a crucial part in ensuring the sustainability of the energy system by reducing the current burden on the energy supply-demand gap. Research on the thermal energy storage concept has been conducted in the last few decades for drying agricultural and food goods using a solar dryer [54, 177]. A change in the material's internal energy, such as sensible heat or latent heat, can be used to store thermal energy. LHS stores thermal energy at the time of phase transition from liquid to vapour and solid to liquid in the form of latent heat of vaporization and fusion respectively

[57]. PCMs have been widely studied and used for a wide range of applications due to their property of heat absorption and discharging during the melting and solidification process respectively [178]. The term "thermal energy storage" (TES) refers to the process of storing energy by cooling, heating, melting, solidifying, or vaporizing a substance. Storage by varying materials' temperature to rise or to lower is known to be Sensible Heat Storage (SHS) [57,58]. Due to the high moisture level and inexpensive price, bentonite clay and CaCl₂ were chosen as desiccant materials for a drier to dehydrate maize [58,59]. Moisture sorption of the 'desiccant materials' is almost 45%(w.b) and also, and they might be renewed at 45^oC. Using solar-collectors hot-air, the dehydration process was carried out during the day. And dehydration process was maintained at night by forcing hot-air from end to end of desiccant layer [179,228]. During the day, solar radiation replaced the saturated desiccant layer. Using this solar drier, 38% of the original moisture concentration level of 90 kg of corn was dropped down to 15% (w.b) in 24Hrs. Similarly, another study looked into the construction of a can-based absorbing solar air heater. The energy efficiency of three different varieties of double-flow solar heat collectors was calculated in different operating situations. Table 5 shows the differences in temperatures between Type-I, Type-II, and Type-III at 0.05 and 0.03 kg/s mass flow rates. At 0.05 kg/s air flow rate, the average temperature difference of air inlet and outlet (Ta) of Type-I, Type-II, and Type-III was determined to be 16.7 $^{\rm O}\text{C},$ 13.01 $^{\rm O}\text{C},$ and 12.27 $^{\rm O}\text{C}$ respectively. The temperatures at inlet and outlet (Ta) differences for Type-I, Type-II, and Type-III were 20.07 °C, 17.19 °C, and 15.14 °C respectively, at 0.03 kg/s flow-rate. At 0.05 kg/s of mass flow rate, the collector efficiency for type I was found to be ideal. The Type I flat-plate collector consistently outperforms the Type II and Type III flat-plate collectors in testing results for efficiency. By generating turbulence and preventing dead zones in the collector, the barriers or cans guarantee that air flows freely over and beneath the absorber plates [60].

The designed system is installed at two positions that are below the dehydration compartment and below the solar-collector [180]. The sensible heat-storage component increased the air thermal efficiency of the solar collector to 28%, while the dehydration chamber's thermal-efficiency was 11.8%. Dehydrating camel meat was used to conduct a performance study. For dehydrating Valeriana Jatatnansi, another researcher used an indirect type sun-dryer with sensible-heat storage with engine oil material and latent-heat storage with paraffin RT- 42 material, with this 89%–9% (w.b) Jatamansi's moisture level is reduced. By integrating the energy-storage materials the dehydration

Several dehydrated agro-products by applying various indirect solar-dyers.

S. No.	Type of Solar Drver	Dried Product	Observations	Remarks	Ref
1	Indirect Solar Dryer:	Straw berries	The dehydration times for the whole, halves, quarters, and discs strawberries were 28, 26, 20, and 24 h respectively	Small farms can use an indirect sun drier, which yields top quality strawberries even in unfavorable weather.	[96]
2	Indirect Solar Dryer	Grapes	At a dehydrating flow of 31.3 m ³ /h.m ² , the dehydration period was 13Hrs.	The thermal transfers, which in turn affect the output temperature and collector performance, are significantly better as compared to the collector without obstructions	[97]
3	Indirect Solar Dryer:	Granny Smith apples	In an indirect type natural convection solar-dryer, Granny Smith apples of 1 kg are dehydrated for 8–10 h at 45.5 °C to 50.5 °C, with a final moisture content of 5%	Small farms can use an indirect solar drier, which produces high- quality products even in adverse weather.	[98]
4	Indirect Solar Dryer	Beans	Whenever beans are dehydrated without any storage elements for 32Hrs and with the integration of storage elements for 24Hrs, the moisture content drops from 65% to 18%.	The findings indicate that the ambient air temperature within the cabinet has the greatest impact on the drying rate. The drying cabinet's internal air speed change has a minor impact that can be disregarded.	[99]
5	Indirect Solar Dryer	potatoes	The moisture content of figs declines from 70% to 20% when dried without storage material for 32Hrs and with storage substance for 12Hrs.	The relative correctness of seven theoretical models was examined and verified using experimental data. The drying curve of potato slices using the two dryers was found to be best fitted by two new dimensionless models.	[100]
6	Indirect Solar Dryer	tomatoes	At a temperature of 65 °C, 0.3 m/s velocity of air, and relative humidity of 30%, the optimum dehydration results for tomatoes are obtained. The	Predicted the kinetics and thermodynamics of tomato drying process.	[101]

moisture level

Table 2 (continued)

	(,			
S. No.	Type of Solar Dryer	Dried Product	Observations	Remarks	Ref
			of tomatoes dropped from 80% to 18% in just two and a half days.		

rate is improving when compared with open-sun dehydration. Therefore, the Jatamansi dehydration processing time is 120Hrs, whereas in the absence of energy-storage materials the dehydration processing time is 216Hrs, while open-sun dehydration took 336Hrs [61]. An indirect type solar-dryer integrated with energy-storage medium of a packed-bed was built, and the design efficiency was evaluated by dehydrating slices of orange. The efficiency of dyer exergy ranged from 50.18% to 66.58% without energy storage and 54.71%–68.37% with energy storage, respectively [62].

An indirect type solar-dryer with a system of forced-convection and an absorbent bed (sensible-heat) packing medium was created and evaluated [63,181]. By drying the Slices of bitter-guard the dryer's effectiveness was estimated under the environmental conditions of Coimbatore. As illustrated in Fig. 6, the dryer had a hot-air solar-collector with a 2 m surface area, fan and dehydrating space. At 0.063 kg/s air-flow, the moisture level of the slices of bitter-gourd dropped from 92% to 9% (w.b) in 7Hrs. At 0.014 kg/s to 0.087 kg/s air-flow, the regular efficiency of exergy changed from 28.74% to 40.67% respectively. Bitter gourd slices had an effective moisture diffusivity of 8.62 x 10^{10} to 12.95 x 10^{10} m²/s with the present setup, whereas with open-sun drying it is having the least value of 0.95 x 10^{10} m²/s. At 0.014–0.087 kg/s mass-flows the normal efficiency of pickup is increased from 17.12% to 54.29% and for exergy, it is increased from 28.27% to 40.68% [64]. Similarly, for crop dehydration, an indirect type solar-dryer with a thermal-energy storage device has been manufactured. The dryer was used to help determine the color and flavor of the produce after continual dehydration [182]. As illustrated in Fig. 7, the dehydrator is developed with a 1.5 m area solar-collector and 6 portions of trays which can hold 12 kg of agro-goods. The dryer had PCM of 50 kg, which is used to preserve 6 °C more dehydrating air-temperature than ambient air-temperature. After sunset, the heat storage device came in handy for completing the dehydration process. As a result, the dehydration period could be greatly reduced [65,66].

4. Natural materials for energy storage in solar dryers

Generally, liquids and solids are capable of storing thermal energy. Substance enthalpy is because of thermo-chemical, latent heat and sensible heat and a combination of all. Various energy-storage systems in solar dryers have been illustrated in Fig. 5 [122,123,183]. Continuously, materials of Sensible Heat Storage (SHS) cause a rise/drop in temperature. These SHS materials are generally used as 'TES' materials in the applications of low/high temperatures due to their comparatively strong thermal stability and heat transmission capabilities [184,226]. Due to their suitability, stability, low cost of investment and effortlessness execution, SHS materials in solid state are employed in a variety of practical applications for storing energy. The topic of naturally available storage materials is covered in this article. Because solid materials have a higher heat capacity, SHS in solid state is a preferable choice for storing thermal energy [185]. Natural materials utilized in SHS can store heat energy without altering phase. Equation (1) governs the amount of solid state thermal energy storage in materials [29-31]:

$$Q = \int_{Tf}^{T} mC_p dT$$
(1)

Dehydrated agro-products under mixed mode and hybrid solar dyers.

S. No.	Type of Solar	Dried Product	Observations	Remarks	Ref
	Dryer				
1	Mixed Mode Dryer	Potato	The drying temperature ranged from 53.8 $^{\circ}$ C to 52.6 $^{\circ}$ C. The air velocity, on the other hand, was set at 0.15 and 0.45 m/s.	For drying potatoes, a mixed-type solar dryer is better, and agro-goods are dried perfectly in a mixed- mode dryer. The thermal storage efficiency is more effective than indirect type dryer	[100]
2	Mixed Type Dryer	Tomatoes	For flat slices, the drying time ranged from 5Hrs to 21.25Hrs. For the wedges, the drying time ranged from 8.5Hrs to 23.25Hrs	Predicted the kinetics and thermodynamics of tomato drying process.	[101]
3	Mixed Type Dryer	Paddy	Solar-dried rice had a higher degree of whiteness than sun-dried rice.	Results showed that quality factors weren't impacted by the solar dryer's operating parameters, except rice's whiteness, which was superior to samples dried in the sun. In comparison to those of the sun drying process, the final product's other quality attributes were all satisfactory.	[102]
4	Mixed Type Dryer	Mango	The moisture content of the mango was decreased to 23% by utilizing mixed-mode dehydrating and by 46.9% when exposed to the open sun for the same amount of time.	This drier is considered to remove moisture from mango during the time of the cold season, and there are different collectors. A combined solar dryer produces high quality dried items.	[82]
5	Hybrid Solar Dryer	Banana	The drier could dry banana slices of 30 kg in 8 h and identified that the moisture level diminished from 82% to 18% (w.b). The color, aroma, and texture of the solar-dried items were superior to those of the sun- dehydrating products	The hybrid Solar Dryering process holds the color, aroma, and texture of the solar-dried items	[83]
6	Hybrid Solar Drier	Onions	Dehydrating an onion from a moisture level of 36%–7% (w.b.) results in a total energy of 23–62 MJ/ke water. By	In a hybrid solar dryer, recommend recycling the hot exhaust air system to save overall energy. The hybrid sun dryer is ideal for drying	[103]

Solar Energy Materials and Solar Cells 255 (2023) 112276

Table 3 (continued)

S. No.	Type of Solar Dryer	Dried Product	Observations	Remarks	Ref
			reutilizing the exhaust hot air, total energy savings of up to 70.7% can be obtained.	onions at a controlled temperature.	
7	Hybrid Solar Dryer	Pineapple	In the biomass, solar, and solar- biomass modes the day-end moisture level efficiency was 11%, 15%, and 13% respectively.	Recommends drying pineapple with a hybrid solar drier and utilizing a hybrid solar dryer in bad weather.	[104]
8	Hybrid Solar Dryer	Tea	A total of 60.2 kWH is required to maintain a temperature of 50 °C in the air.	At low temperatures air, the hybrid sun dryer is recommended.	[105]

Where.

 $\begin{array}{l} T_i = Iniital \; Temperature(K) \\ T_f = Final \; Temperatures(K) \\ m = Materials \; Mass \; (kg) \\ C_p = Specific \; Heat(kJ/kg.k) \\ dT = Change \; in \; Temperature \end{array}$

Since the non-boiling and freezing nature the following materials are used as soil, bricks, concrete, clay, gravel, sand and rocks are solid state natural energy storage for the applications of high/low temperatures [189–192]. These natural materials cannot be affected by the limitations of fluids and superheated steam. The flow rate, heat-transfer rate, packing density and size all have a role to enhance the efficiency of energy storage [122–124]. Some of the readily available natural energy storage materials include reinforced concrete, quartz, bricks, soil, clay, limestone, pebbles, rocks, gravel, sandstone, sand etc. Various natural materials for energy storage are used in different solar applications like solar water-heating, solar air-heating, solar drying etc., by integrating several storage strategies throughout the previous two decades. The development of solar-dryers that are combined with natural materials for energy storage is shown in Table 6 [125]. The results of the study revealed that sustainable methods were introduced for a variety of thermal-energy storing processes [126]. Environmental impact examination on indirect type solar dryers (ISD) coupled to a porous-bed SHS material yielded an estimated energy payback period of 2.2 years [127]. And it is claimed that the forced-convection solar dryer (FCSD) produced hygienic copra slices [128–130].

NES materials, which supply efficient and ecologically friendly energy during sun heating applications, can significantly meet society's needs. They ensure certain significant advantages over thermo-chemical energy storage systems [131,132], like low CO_2 emissions, enhanced ambient temperature, low CFC emissions, low cost for space cooling and heating, low electricity charges etc. NES materials like clay, gravel, soil and rocks may store up to 108 MJ/m³ vol. of thermal energy [133,193]. Since these materials have exceptional thermal stability, they do not lose weight or degrade at high temperatures >100 O C, resulting in a lengthy life span. Due to stratification being conserved over extended periods, the energy storing material cost is insignificant, and solutions underground have some benefits so, the ground is used as insulation. Their utilization allows solutions efficiently over large scale applications through Underground Thermal Energy Storage (UTES) systems. A system which is integrated with energy storage and solar-dryer is mostly

Advantages and disadvantages of solar dryers.

S. No.	Type of Dryer	Advantage	Disadvantage
1	Natural or Direct Solar Drying	 Simple construction, Loading and unloading are straightforward, The dryer is rain and dust resistant, and Agro-products can be dried at night or in the rain. There is less product contamination as a result of the enclosure's clear lid. Provide shelter from the rain, dew, and other elements. The acquired product quality is superior to drying outdoors in the sun. 	 Crop damage is caused by rodents, birds, and other animals. Degradation is brought on by exposure to the sun's direct rays as well as to rain, storms, and dampness. Pollution from the environment, wind-borne debris, and dirt and dust contamination Damage from excessive drying Insect infestations development of microbes. Additional losses during storage as a result of inadequate or uneven drying
2	Indirect Solar Drying	 The rate of dehydration is faster when using a direct dryer. Slightly elevated dehydrating commodities, and conserved flat plate solar collectors were used and gave an excellent outcome, dehydrating performance, It is also appropriate for low or small-scale farms. This method prevents contamination of the finished product. It is a far more effective method of solar drying than the direct form. Prevent direct solar radiation exposure to preserve product quality. Some products dry more quickly than others. Final product conditions are not based on the conditions of a natural phenomena 	 drying Construction costs are high. After a certain time, maintenance is required.
3	Mixed mode sun drying	 Accelerated drying with safe product moisture levels. Compared to other drying methods, drying takes less time. 	 Agro-food capacity is very less. Grain dried over a year has a worse quality than grain dried using an indirect form of dryer. The cost of maintenance. The required capital cost is higher.

beneficial for allowing the drying process continuously even after sunset, with this the dehydrating time is significantly minimized [134]. SHS or LHS systems can collect and store solar thermal-energy. As shown in Fig. 5, two types of energy storage technologies are now being explored and developed in solar dryers and the evolution of NES materials in different solar-dryers is shown in Table 6. The NES materials are positioned in a solar air collector (SAC) below the absorber plate in the first configuration [188,193]. The absorber sheet is in direct contact with it, and the absorbed thermal energy is delivered to 'TES' system via conduction mode of heat-transfer. The NES materials are retained in a heat exchanger of box type among the SAC and the second type is having a drying chamber [135–137]. The heat exchanger collects and stores energy, which is then utilized for drying products with incoming hot-air shown in Fig. 8. The NES system is located under the drying chamber

in the third design [138]. The charged air from the SAC passes via the TES structure; there it is captivated by the NES materials and used for drying purposes during the hours when the sun is not shining [139,140].

5. Performance indicators for solar dryers

To evaluate the performance of solar dryers, numerous methods and processes were developed based on many merits, such as energy efficiency, time to dry and product quality. In solar drying, thermal performance is a reliable indicator to study the system's merits and can be quantified using energy analysis. Energetic performance is based on the first law of thermodynamics, which takes in to account the quantity of energy and the energy change with respect to the change in surroundings [144,187]. However, the drawback of energy analysis is that it only considers energies at the inlet and outlet of the system, and sometimes is redeemed as insufficient for system optimization as it neglects the irreversibility and thermodynamic losses [145-148]. In general, energetic analysis on solar dryers can be done on two main components: the drying systems and the drying materials. Drying systems of solar dryers include the solar absorber unit, drying chamber, and movement of heated drying air throughout the system. In short, energy analysis of solar dryer components is commonly done by applying heat transfer and energy balance based on the principle of energy conservation of the first law of thermodynamics. Determination of the thermal performance of solar dryers is important to achieve maximum moisture removal while using minimum amount of energy [149,227].

In literature, several indicators are commonly used to evaluate the thermal capacity of solar dryer components, especially for solar collector units. The amount of useful heat that can be harnessed from solar collectors can be calculated using heat removal factor (F_R) and incident solar radiation (I_t). Q_u value is depended on the material of construction used for collector, as well as the surface area, as suggested by Eq. (2) [150].

$$Q_{u} = F_{R}A_{C}[I_{t}(ra) - U_{L}(T_{t} - T_{a})]$$
(2)

The energy used for moisture evaporation can be calculated as [151].

$$E_{vap} = M_{water} X H_{fg}$$
(3)

Thermal efficiency of a solar collector is the ratio of heat gain by air passing through the collector to the energy gained due to solar irradiation, given by Refs. [152,153].

$$\eta_{\rm C} = \frac{m_{\rm c}(T_{\rm out} - T_{\rm in})}{A_{\rm c}I} \times 100\%$$
(4)

Another indicator commonly used in energetic analysis is the thermal efficiency of solar dryers η_d . Essentially η_d is the ratio of energy required to evaporate product's moisture to the energy consumed for the drying process. In short, thermal efficiency of the drying system is the ratio of the energy used for moisture evaporation to the energy input to the drying system.

$$\eta_{d} = \frac{E_{evap}}{E_{input}}$$
(5)

In passive convection dryers, dryer efficiency calculation is based on the air movement due to natural buoyancy, whereas active dryers take into account the energy input through electrical fans or blowers, given by respectively [154,155]. Depending on the type of solar drying system, the energy consumed for the drying process would need to account for all sources of energy generated in the system. In hybrid systems, usually photovoltaic-thermal (PVT) hybrid dryers, the electrical efficiency of the solar collector is quantified as the system takes electricity into energy generation.

$$\eta_{d,P} = \frac{mL_{v}}{A_{c}It_{d}}$$
(6)





(b) Phase Change Materials

Fig. 5. Materials for thermal energy storage and Phase Change Materials.

$$\eta_{d,A} = \frac{mL_v}{A_c It + P_f}$$
(7)

The relationship between energy input to solar dryer and amount of water evaporated can also be used to define the performance of the dryer and to compare performance of the dryers, is Specific moisture extraction rate (SMER) in kg kWh⁻¹ relates how much moisture can be removed per unit of energy, whereas specific energy consumption (SEC) is the reciprocal of SMER with units of kWh ^{kg-1} [156]

$$SMER = \frac{Amount of moisture evaporated}{Energy input to the dryer}$$
(8)

$$SEC = \frac{Energy input to the dryer}{Amount of moisture evaporated}$$
(9)

Pickup efficiency, or moisture removing efficiency of drying air is the

efficiency measure of moisture extraction using hot air, and it can be calculated using

$$\eta = \frac{h_o - h_i}{h_{as} - h_i} = \frac{W}{\nu \rho(h_{as} - h_i)}$$
(10)

In hybrid systems where energy source comes from other than solar energy, solar fraction is determined to quantify the ratio of energy extraction of heat from solar collector to the overall energy available for the drying process [151]. Solar fraction can be expressed by

$$SF = \frac{Q_S}{Q_t} = \frac{\text{Heat gain at collector}}{\text{Total heat supplied to dryer}}$$
(11)

From the drying material components, effectiveness of drying can be associated with moisture reduction within the samples. The mass of water removed (W) from a wet product can be calculated by [149].

Solar collectors of Type-I, Type-II, and Type-III temperature differences in various air-flows [60].

m (kg/s)	Туре	I (W/m ²)	Time (Hrs)	T _a in ^o C	T1 °C	T2 °C	T3 °C	T4 °C	T _a Out ^o C
0.03	III	420	9:00	23	25	27	26	28	29
	III	950	13:00	40	46	49	56.5	59	64
	III	650	16:00	30	32	35	36	33	39
	II	420	9:00	23	25	27	27	27	29
	II	900	13:00	40	42	60.5	54	48.5	64
	II	650	16:00	30	33	41	39	35	44
	I	480	9:00	23	24	26	26	28	32
	I	950	13:00	39	43	46	52	59.5	67.5
	I	690	16:00	32	33	35	39	44	48
0.05	III	450	9:00	23	24	26	27	26	28
	III	940	13:00	40	42	45.5	48	53	57.5
	III	660	16:00	29	31	33	36	37	38
	II	420	9:00	23.5	24	26.5	26.5	27	27.5
	II	900	13:00	39	44	46	50	54	57
	II	650	16:00	31	32.5	35	37	38	41
	I	450	9:00	23	24	25	28	29	30
	I	990	13:00	39	44.5	49	58	61	65
	I	680	16:00	31	32	34	37	45	42.5



Fig. 6. Various instruments and configurations of bitter gourd slices were used in an experimental setup [63].

$$W = \frac{M_0 (M_i - M_f)}{100 - M_i}$$
(12)

Moisture ratio, which is a dimensionless form of moisture content, explains the ratio of remaining moisture to be removed at time t over initial total moisture present. In the study of drying, MR is an important tool to understand the kinetics and drying profile as they vary from one material to another. MR is found to be mostly adequate to describe the diving behavior of some fruits and vegetables as it translates to drying constant, k (s⁻¹). This is an important parameter widely used in thin-layer modelling, to obtain drying curves as a function of time [157].

$$MR = \frac{(M - M_e)}{(M_o - M_e)}$$
(13)

TES is one of the most practised technologies to store energy in the form of heat to eliminate the gap between the energy supply and demand. As shown in Fig. 5, there are two main thermal energy storage technologies: sensible heat storage through a temperature change (sensible heat) of a material, latent heat storage through phase change (latent heat) of a material and thermochemical heat (chemical energy) by thermally induced changes in materials' chemical states. As compared in Table 7, the choice of TES method depends on a variety of factors such as the storage capacity, cost, temperature range, duration requirement as well as the specific application.

5.1. Pros and cons of sensible heat storage

Sensible heat storage materials are typically based on relatively low cost materials and thus extensively used, except the liquid metals. Due to the relatively good thermal stability, heat transfer performance and transport properties, sensible heat storage materials are the most used TES materials for high-temperature applications. Compared to latent heat storage, the specific heat of sensible heat storage materials is 50–100 times smaller, leading to the requirement of large volumes or quantities to deliver the amount of energy storage necessary for high temperature thermal energy storage applications. The other main issue of sensible heat storage is that the temperature of the storage medium decreases during discharging process, so the heat transfer fluid temperature also decreases with time.



CUT VIEW Fig. 7. A flat plate absorber with thermal storage [35].

5.2. Pros and cons of latent heat storage

Latent heat storage with PCM has a large latent heat of fusion so it can store more amount of heat than sensible heat storage. This large difference gives PCMs the advantage of a high energy storage density, which reduces the volume of the TES vessel and the outer wall surface area, and minimizes heat loss. Compared to sensible heat storage, the discharging process of the latent heat storage can maintain the temperature constant and make the contacting medium like HTF or the adjacent space temperature stable. However, the main drawback of latent heat storage is the low thermal conductivity of PCMs. Moreover, the heat transfer design and media selection are more difficult, and the experience with low temperature PCMs like inorganic salts has shown that the performance of the materials can degrade after moderate number of freezing-melting cycles.

5.3. Pros and cons of thermochemical heat storage

The thermochemical heat storage system is unique and suitable for solar energy storage owing to its advantages: high volumetric storage density, low volume requirement, long energy preservation duration periods with limited heat loss, low storage temperature (ambient temperature) and unlimited transport distance. However, it also has some issues to be tackled, such as poor reactivity and reversibility of reactions, harsh reaction conditions, toxic and corrosive products from reactions, etc. Currently, most studies are conducted at laboratory scales. Largescale tests are needed to verify the feasibility and durability of longterm thermal energy storage. Furthermore, the criteria of material selections for thermochemical heat storage should be established.

6. Quality, economic, social and environmental impact of solar dried products

Due to the wide availability of solar drying systems in practice, the

development of solar dryer technologies needs to be based on empirical knowledge of its energy profile and the anticipated performance over its expected lifetime. The information acquired from the literature is to identify and improve the quality of the products, plant operation costs, energy conservations, fuel versatility and pollutants [194]. In addition, selection of the right dryers must take into consideration the user's need and the end use of dried products, thus requiring the evaluation of the following domains. The impact of solar dried products on quality of the dried products, the economics of the dried products, social impact and environmental issues are studied for a better understanding of the solar energy conversion for post-harvesting process. Due to the widespread use of sun drying systems in daily life, the advancement of solar dryer methods must take their energy profile and predicted performance over their estimated lifetimes into consideration. Literature-based knowledge is used to evaluate and enhance product quality, plant operating costs, energy efficiency and emissions. The user's needs and the intended use of the dried goods must also be taken into account when choosing the appropriate dryers, necessitating an assessment of the following domains.

6.1. Quality assessment parameters

Physical, biological, chemical and nutritional quality qualities of dried agricultural produce can be classified. These quality factors play a big role in dried product specification. In general, improper dehydration processing conditions lead to significant nutritional loss and material degradation [67]. Although dehydrated materials are a good nutrient source, the dryness procedure may alter the quality. Color, flavor, and nutrient preservation are all quality characteristics. The market value of food is determined by these factors. Pretreatments and novel dehydration processes can aid in the production of good-quality food [68–70]. The color factor is an important qualitative attribute that determines customer approval. The total color changes of food during dehydration are influenced by color and enzymatic browning [71,72]. At high

Evolution of NES materials in different solar-dryers.

S. No	Type of Dryer	Dried Product	Energy Material	Observations	Ref
1	Indirect Solar Dryer	Fruits and Vegetables	Sand	When compared to drying time without storage material, the TES system with a dryer reduced drying time by 12Hrs.	[141]
2	Indirect Solar Dryer Forced Convection Solar Drier	Copra	Sand	A dryer was used to dry the copra integrated with sand. In the bottom and top trays, moisture content was reduced from 51.78% to 7.77% and 9.63% after 32Hrs. When compared to a dryer without sand material, the drying rate was faster	[136]
3	Solar Green House Dryer	Coconuts	Sand	In 73Hrs, the drier concrete as a natural energy storage component and reduced the moisture content from 52% to 7%. The OSD took 174Hrs to complete	[142]
4	Indirect Solar Dryer	Copra	Sand	For SAH with and without energy storage components, the specific moisture removal rate (SMRR) was calculated to be 0.81 and 0.94 kg/ kWh respectively	[128]
5	Flat Plate Solar Air-Heater	Agro- Products	Sand	According to reports, the thermal qualities of air heaters made of SHS materials were the best.	[129]
6	Solar Air- Heater	Coriander	Rock Bed	By integrating rock- bed, the coriander is dried for more than 13Hrs and the moisture level is lowered from 73% to 53%. Without a rock-bed, the drying process is completed in 3- days.	[137]
7	Indirect Solar Dryer Forced Convection Solar Dryer	Bitter Gourd	Pebbels	A maximum drying rate of 2.3 kg of moisture is removed with pebbles	[127]
8	Indirect Solar Dryer	Camel Meat	Pebbels	The drying time has been slashed by 10Hrs.	[143]
9	Solar Dryer integrated with the packed bed	orange slices	Pebbels	Die drying system exergy efficiency increased from 54.7% to 68.4% while using TES.	[126]
10	FCSD combined with gravel	Chillies	Gravel	The use of NES material increased the dehydrating time by 4H/day	[144]

temperatures, the process of dehydration is sped up by increasing the air's moisture-holding capacity. However, depending on fresh product thermal-sensitivity, high temperatures exceeding 80 °C damage the color and texture of fruits and vegetables [73].

Between 50 °C and 30 °C, the effect of dehydrating air temperature on dry chilli has been investigated [74]. In this research, the dehydration period was shortened, and the effective moisture diffusivity was increased, thanks to the higher temperatures. High air temperatures, on the other hand, had a significant impact on chilli quality, resulting in losses in volatile chemicals, nutrients, and color factors. Several food aromatic chemicals influence taste in the flavour of foodstuff. Several taste components become volatile when moisture is removed using the dehydration process, which needs to be controlled. The food microstructure has a significant impact on the texture and shape that affect flavor expression during the dehydration process. Chemical analyses or sensory evaluations can be used to investigate a food material's taste components. Organic analyses can deliver quantitative information on fragrant compounds, but they are invalid in terms of human taste perception. As a result, sensory evaluation plays an important part in determining whether or not food products are ultimately acceptable [71].

Industrialization of an indirect solar dryer for drying apples and watermelon slices [75]. The apple's initial moisture content dropped from 6.16 to 0.799 kg. Under the indirect sun drier, the watermelon's moisture level dropped from 10.76 kg to 0.496 kg. The color change of the watermelon and apple dried in the open sun was different for the watermelon and apple dehydrated in the indirect type solar dryer as shown in Fig. 9. The drying kinetics of turmeric is studied using direct and convection solar dryers. The direct sun dryer reduced the browning index of the samples by 25.65%, whereas hot air convection drying decreased it by 21.53%. In the direct sun dryer, the spherical particles examined with electron microscopies were more exposed and smoother, showing that the flow ability spots had been improved through packing processing [76]. The physical properties of the product are frequently affected by drying, resulting in variations in texture, color, shape and size. During dehydration, many enzymatic and chemical reactions will take place. Even though not all of these phase changes are worse, only a few of them turn the product into taste bad. It's important to compare the quality of solar dryers because it might vary greatly based on the air-flow rate, dryer design and drying temperature. Sensory and nutritional factors, as well as rehydration capabilities, are frequently assessed as part of a dried product's quality assessment [77-79]. Table 8 summarizes the attributes used for assessing the quality of the dried product.

6.2. Economic assessment

Evaluating the solar dryer methodologies for technical and financial viability involves economic and financial analysis. The solar-dryers are recognized as capital-intensive compared to other industrial dryers [106,167]. Solar dryers also provide various financial advantages over traditional sun drying processes, such as cheaper expenses for combustion equipment and fossil fuel materials. They also provide high-quality outputs, increasing market value and ensuring steady and high revenue despite varying climatic circumstances [186]. Solar dryers, in particular, allow small-scale farmers to minimize post-harvest wastage in an energy-saving and cost-effective manner, and also improved quality of products may propose new ways for employment and income opportunities. Payback period, life cycle savings and annualized expenses have all been used to assess the economic possibilities of solar-dryers in the literature [107,168]. However, it should be emphasized that numerous factors influence these calculations, including dryer system, climate conditions and physical qualities of the dehydrating materials [108, 109]. Solar drying, for example, is economically viable for cash-type crops like coffee and tea in India, but unfortunately not economically viable for the vegetables like tomatoes and cabbage. The anticipated payback period for a 1000 kg volume greenhouse type drier for fresh



Fig. 8. Solar drier used for copra drying [128].

Table 7
Available energy storage materials used in the thermal energy storage systems [164,165].

Material	Туре	Density (kg/m ³)	Thermal conductivity (W/m-K)	Heat capacity (kJ/kgK)	Cost (€/m ³)
Rock	Solid	1500-2800	0.85–3.5	1	64–742
Concrete	Solid	2000	1.35	1	76
Sand and gravel	Solid	1700-2200	2	0.910-1.180	6–8
Ceramic tile	Solid	2000	1	0.8	1600-3500
Gypsum (coating)	Solid	1000	0.4	1	78
Ceramic brick	Solid	1800	0.73	0.92	36-64
Wood	Solid	450	0.12	1.6	404
Water	Liquid	990	0.63	4.19	1.6
Oil	Liquid	888	0.14	1.88	6560
Nitrite salts	Liquid	1825	0.57	1.5	2200
Carbonate salts	Liquid	2100	2	1.8	6050
Liquid sodium	Liquid	850	71	1.3	2000





Indirect Solar Dried Apples



Fig. 9. Drying of apple and watermelon slices in IDSD and OSD [75].

items like chilli in Thailand is roughly 2 years [110,169]. In general, projections for solar dryer technologies' economic returns priced at market prices and ignoring any potential good environmental benefits, are substantially greater than traditional drying technologies. Other benefits, like low land use, larger yields and greater quality should be addressed to provide a more realistic cost-benefit analysis [195–198].

While energy analysis is a common approach used to minimize

thermodynamic efficiencies within dryer systems, thermo-economic is a different take to estimating the cost-optimal structure and the optimal values of thermodynamic efficiencies in each component [158]. Thermo-economic is viewed as a promising diagnostic tool, even for complex systems [147]. Through economic analysis, solar dryer application has been proven to have an undeniable improvement in carbon footprint reduction associated with the energy-intensive drying process.

Different attributes for measuring the quality of the dried product.

S. No	Attributes	Observations	Type of attributes	Remarks	Ref
1	Sensory attributes	Sensory assessment for quality is a mixture of multiple perception senses that comes into play while deciding what to eat and how much to eat	Color, size, shape, homogeneity	Chemically, taste attributes may be approximated, but their optimum in terms of customer choice differs	[80]
2	Nutritional attributes	Organic characteristics are analyzed	Sugar and ash level, Vitamin-C and ascorbic acid level, acidity level and beta- carotene level	High amount of acidity level indicates degraded quality because of fermentation	[80]
3	Rehydration attributes	Rehydration capacity describes the amount to which it can restore lost moisture.	appearance, texture, aroma and flavor	The rehydration capacity of dried fruits is discovered to be influenced by cell wall, intercellular tissue and pectin component	[80] [81] [82]

Solar dryers are highly effective devices with low investment to produce good quality dried products. The unit costs of useful energy for solar dryers were found to vary from 0.0034 to 0.015 USD per MJ of energy for different types of drying products [159]. An economic study to evaluate monetary savings due to the application of industrial solar dryers under the Lebanese climate indicates that the energy cost saving is determined on monthly basis, where it is dependent on the percentage of time when the solar dryer is used (P_r) the dryer energy consumption for operation E_{month} and the cost of electricity for one unit of kWh, P_{kWh}, depending on the P_r value which ranges from 0.1 to 1, the energy cost reduction records savings between \$130 to \$4160 per month for drying of 120 kg of various vegetable samples.

$$SM = P_r \quad X \quad E_{month} \quad X \quad P_{kwh}$$
 (14)

From the determined SM and capital cost of the solar dryer, simple payback period for the dryer system was determined as follows

$$PP = \frac{C_{dryer}}{SM}$$
(15)

Another approach for economic analysis is the incorporation of costbenefit analysis to compare the cost and benefits of solar drying to other means by taking into consideration the size, materials for construction, efficiency, operation, sophistication and sustainability of the driers which vary from country to country. Table 9 presents a comparison of economic analysis methods applicable to solar drying systems.

6.3. Social assessment

Global warming poses a huge danger to the world, especially because of high GHG emissions levels. Transferring to solar based energy from fossil-fuel energy has been attempted by governments and enterprises all around the world [111,199]. Most governments deal with the problem by enacting energy policies that include laws, international treaties, and

Table 9

Comparison of economic analysis methods for solar drying systems.

S. No	Solar Dryer	Drying material	Economic Indicator	Findings	Ref
1	Mixed mode greenhouse SD	Red pepper and grape	Annualized cost of dryer Annualized capital cost (Cac) Annual electricity cost for Annual savings (Sj) for drying the typical product in the jth year Payback period	Dryer capital cost rs 660 USD. and Payback period is 1.6 years compared to 20 years of lifetime.	[160]
2	Modified greenhouse dryer	Potato chips	Payback period	Payback period is 1.11 years.	[155]
3	Indirect cabinet SD	Carrot, Corn, Mushrooms, Potatoes, Apples. Banana Cherries. Peaches	Amount of saved money Payback period (PP)	The capital cost of dryer is 8000 USD. and savings recorded range from 1400 USD to 12500 USD depending on mass and type of drying sample and percentage of dryer utilization. From this, the payback period range from 0.9 to 62 months.	[161]
4	Mixed mode SD	Various agri- produce	Capital cost of dryer Unit cost of drying Unit cost of useful energy Valuation of benefits		[162]
5	Low cost SD -Direct and indirect passive dryers	Fish	Fixed cost - construction and maintenance cost Qualitative performance evaluation		[163]
6	Direct and indirect passive dryers	Agro Products	Carbon savings	Life-cycle assessment on environmental performance of industrial solar thermal system (ISTS). Large- scale ISTS applications were found to achieve energy and carbon savings ranging from 35 to 75 GJ and 2–5 tonnes of C02 per kWth. depending on the geographical location.	[166]

financial incentives for investors [112,113]. Most countries like Malaysia, China, Spain Canada, Germany and France are supporting and implementing clean and renewable energy for various applications. Renewable energy sources accounted for most of the global energy depletion which is 19%, with solar, geothermal, biomass and wind [114, 200]. Almost 9.8 million jobs are available in the renewable energy sector. The majority of possible jobs are from Asia, especially from China [115]. Furthermore, solar-energy is considerably having benefit to public health and is eco-friendly, since several studies have discovered that greenhouse gas emissions (GHG) can increase the risk of allergies, cancer, cardiovascular morbidity and non-allergic respiratory disorders [116,201].

6.4. Environmental assessment

Most drying systems use electricity and fossil fuels as sources of energy, resulting in high operating costs and environmental issues due to increased GHG emissions. Because of this reason, farmers and scientists have turned their focus to clean energy technologies like solar and thermal energy, which can be used directly or indirectly. The consumed energy of the solar dryer is advised to be calculated with help of indicators like carbon mitigation, CO₂ emission and embodied energy [116,117]. It has been discovered that a solar dryer system with a 40% efficiency level can reduce conventional energy usage by 27%-80% [118,202]. The fan consumes 5% of the total energy used in the forced ventilation greenhouse drier. Conventional and Solar energy collectively can save energy up to 20%–40% [119]. By utilizing electrical energy for 275 days at 100 kWH per day, the drying system with hot air produces carbon dioxide of around 15 tonnes per year [120,181]. Finally, it was determined that using solar dryers possibly will diminish emissions of CO₂ when related to general non-conventional drying methods [121].

7. Policies, challenges, risks and recommendations

7.1. Polices

Recently, solar applications have significantly been considered due to the importance of protecting environment by governments. Government policies can reduce the energy crisis and the effects of global warming. Governments can provide great interest and motivation in using solar technologies, which include solar drying systems by applying policies such as renewable portfolio standards (RPS), feed-in-tariff (FIT), and incentives [204]. Some researchers also studied the solar energy related policies that were implemented in various countries including the united states, Germany, Spain, China, Australia, Pakistan, France, Malaysia, and Canada. Most of these policies deal with environmental issues and using energy efficiently [205]. Globally, numerous policies have been created and put into place to encourage the use of renewable energy for various processes apart from generating electricity alone, including feed-in tariffs (FIT), renewable portfolio standards (RPS), tax credits, pricing legislation, production incentives, quota restrictions, trading systems etc. (RE) [206]. Primary goals of these include minimizing the negative effects of energy and its allied sectors on the environment, decreasing dependency on fossil fuels, and promoting new industrial development [207,208]. However, the feed-in tariff (FIT) and the renewable portfolio standard (RPS) are the most widely implemented policies. Despite the numerous arguments around their efficacy, in general, a decision must be taken amongst them. In this case, the countries could choose which RE policy is appropriate given their unique situations and goals [209].

Industry statistics state that worldwide and in the United States, solar energy deployment grew at a record rate in 2008. The "2008 U S. Solar Industry Year in Review" report from the Solar Energy Industries Association revealed that the country's solar energy capacity expanded by 17% in 2007 to reach a total equivalent of 8775 MW (MW). In 2007, the United States installed 342 MW of solar photovoltaic (PV) electric generation according to a report by the Solar Energy Industry Association (SEIA) [210,211]. As of 2010, 12% of the country's electricity came from all renewable sources (solar, wind, geothermal, hydroelectric, biomass, and waste). By 2030, the DoE (Department of Electricity) wants to produce 10–15% of the country's energy from solar sources [212]. By 2025, solar energy might supply up to 10% of the country's energy requirements. According to a forecast created by the organization Co-op America and the research and publishing company Clean Edge, solar photovoltaic systems will produce more than 8% of the country's electricity, while concentrating solar power systems would produce close to 2% [213].

Renewable portfolio standard: In its 28 states, the United States of America has implemented the RPS mechanism, which calls for the use of renewable energy sources including solar, wind, geothermal, and biomass for the maximum amount of production energy. Most policy goals seek to encourage the diversity of power-generating sources, lessen reliance on fossil fuels by the state, boost the use of renewable energy, cut carbon emissions, or other combinations of these goals [214, 215]. For the most part, jurisdictions with RPS regulations permit utilities to trade renewable energy certificates (RECs) to meet their RE responsibilities.

Feed-in tariffs: Feed-in tariffs (FITs) have proven as the most successful government incentive programme for renewable technologies. In reality, FITs are responsible for half of all PV installations worldwide [216]. Around the world, FITs for PV are in use; in early 2009, FITs were available in 45 nations and 18 Canadian states, provinces, and territories. The paper creates the FITs system in Germany with a 20-year contract period and consistent payment for the energy produced. For the various types and rated powers of the generation system, different values of FITs are established. In late 2008, it was agreed to revise the Erneuerbare-Energien-Gesetz (EEG) to encourage a stronger price decrease. For new PV systems, a rate of FIT values depreciation was chosen [217]. FITs for electricity produced in Spain from PV systems, according to Real Decreto 436/2004. One of the primary policies for PV in France is the FIT. This system was released in July 2006. It entails a utility's responsibility to buy electricity produced by renewable energy producers in its service area at a price set by public authorities and secured for a set amount of time. The value of a FIT is the total amount paid to an independent producer for each kWh of electricity generated by a RES-based system, including any premium over or above the market price but excluding any tax credits or other government-funded production subsidies. In both Europe and the USA, FITs have been the primary method for assisting RES development. They are now being used in 20 EU member countries [218]. While various Australian governments and territories have looked into FITs, only Queensland and South Australia now have programmes in place [219,220]. Queensland's Solar Bonus Scheme rewards small power users, who are those that use less than 100 MWh per year. AU\$ 0.44 kWh⁻¹ from photovoltaic systems less than 10 kVA for single phase and 30 kVA for 3-phase systems until 2028 [221,222]. Similar rules apply in South Australia, although the small electricity user must use less than 160 MWhyr⁻¹.

Government incentives: Government has a significant role to play in slowing the trend of GHG emissions. The government may enhance the adoption of PV through thoughtful policy initiatives, which will encourage related innovation and boost economic competitiveness through economies of scale. Governments of any size can lessen the environmental impact of their regions through a reduction in GHG emissions from carbon usage by increasing their reliance on distributed sources of renewable energy, especially roof-mounted PV. Overall, it is clear that the government is interested in expanding distributed renewable energy production in its regions, such as roof-mounted solar power [223]. Credit terms and tax incentives favour additional subsidies. They vary depending on whether the PV installations are carried out by an individual or a business investor. Commercial systems are exempt from VAT (value-added tax; 19% VAT in Germany). Additionally, some German areas offer investment grants.

Renewable energy law: The Law on Renewable Energy was passed by the National People's Congress in 2005 [224]. The growth of renewable energy in China has entered a new phase thanks to this regulation. Since REL was first introduced, numerous enabling laws and directives have been adopted to put it into practice. Guidelines for Using the Public Fund for Renewable Energy Development were published in 2006 by the NDRC (National Development and Reform Committee). Three priorities are listed in these rules. One is promoting the use of renewable sources of electricity (including wind, solar, and ocean). The other two goals are assistance with the use of renewable energy in building heating and cooling systems and research into energy alternatives that can replace oil. They agreed that there are two ways the public fund can be utilized. It may first be given out as a grant. Such grants are used by the recipients to conduct research and development on renewable energy sources. It can also be used to pay the interest on loans. Renewable energy projects that qualify may receive public funding to cover some of their loan interest.

7.2. Challenges

The design, construction, operation and examination of solar dryer systems as well as projects associated with them involve several issues, including structure [203–206], Political and social issues [229,230], economic provision, marketing, environmental condition [231,232],

knowledge and efficiency [233]. Six potential criteria are examined, and the proposed factors are briefly described in Fig. 10. The quality and rate of drying of agricultural goods are efficiency factors. Geographical conditions relate to solar radiation quality and the use of solar dryers as a means of preventing climate change, as well as infrastructure factors that include things like infrastructures, the potential for luring private investors, the expertise of local engineers, economic sanctions, and the financial advantages of solar dryer use [225]. The shortcomings also include a lack of adequate production experience, a lack of executive management experience for the implementation of solar dryer projects, a lack of competition in the design, production, and supply of solar dryers, as well as other factors like loan borrowing and government financial support for related projects. Finally, the social, cultural and political aspects covered in the suggested questionnaire include farmer knowledge of solar dryers, indigenous people's choices about foreign-made dryers and their readiness to adopt the technology, as well as political hazards.

The quality of dried goods produced by solar dryers is superior to that of goods produced by conventional dryers. In sun dryers, product drying occurs quickly. A benefit of erecting solar dryers is adequate sun radiation. It is a good idea to use energy resources in solar dryers to stop climate change. Building solar dryers involve infrastructure (such as the required skills, tools, and so forth). To install solar dryers, it is feasible to entice private investors. Engineers need the appropriate tools, skills, and



Fig. 10. Factors affecting the Solar Dryers Progress.

resources to create solar dryers. The main obstacle to the use of solar dryers, notwithstanding their low cost, is economic sanction. The government is giving solar projects significant funding. Farmers are not sufficiently informed about solar dryers and their advantages.

7.3. Risks

In addition to all the significant project-related considerations, several project-related dangers could interfere with the effective completion of associated activities. Therefore, in addition to the associated components, three distinct types of the aforementioned risks—financial, external, and building risks—have also been analyzed [234]. Fig. 11 presents the risk-related factors that were previously mentioned. According to the figure, budget and pricing risks, inflation, interest rates, currency rates, and executive risks are included in economic risks. Structural risks include construction hazards which take into account discrepancies between designing and building, environmental and safety issues and risks associated with project execution. Finally, external risks consist of economics and environment, laws, political risks and demand concerns [235].

7.4. Recommendations

Authorities must pay more attention to the characteristics that have been found and look into how they affect the advancement of projects related to solar dryers to increase the application of these systems. The elements and risks impacting the accomplishment or failure of connected projects in the world have been identified in this study. Therefore, it is advised that the influencing elements suggested in this study be given more consideration by governments and businesses. The results show that to decrease the effectiveness of these solar plants, the government and businesses must concentrate on the important elements affecting solar dryers. Solar dryers with higher efficiency would be built and manufactured by raising and updating the level of engineers' expertise and information and developing the designing and producing procedures. The results show that to decrease the effectiveness of these solar plants, the government and businesses must concentrate on the important elements affecting solar dryers. Solar dryers with higher efficiency would be built and manufactured by raising and updating the level of engineers' expertise and information and developing the designing and producing procedures.

More investors would be enticed to invest in solar drying systems if

more people, including consumers, managers, and officials, were aware of their cost-effectiveness. On the other hand, as more private investors are drawn to projects using solar dryers, more attention will be paid to the fundamental framework for their execution, including the acquisition of the most recent information, design, and manufacturing technology. In the end, this will boost the effectiveness of solar drying systems. Therefore, one strategy that might be suggested in this situation is to educate and increase public awareness of the advantages of using solar dryers, particularly among farmers. Farmers' willingness to accept solar dryers will rise as their knowledge of them rises. As a result, it was easier to promote these systems to both niche and broad consumers. Furthermore, the government should be aware of this problem to further develop solar dryers as interventions have a direct impact on their implementation. The likelihood of completing solar dryer-related projects will rise as more expertise is gained in their design, building, operation, and management. However, as competition in this field grows, officials and staff members' efforts to enhance these systems' performance will also grow.

8. Conclusions

A detailed review of recent studies on solar dryer systems involving natural energy materials and storage systems has been presented. A review of traditional solar drying procedures and their improved versions, and an examination of research needs based on various technoeconomic-socio-enviro-policy factors have been consolidated. The necessity to produce higher quality foods in large quantities is currently pushing the development of innovative dehydration technologies. This goal can be achieved by integrating solar dryers with natural energy storage materials to produce high-grade dehydrated food-stuff at a low cost. Agro-Products pre-treatment is very much required before the drying process to diminish the ultimate food quality deterioration. The combined dryer setup with PCM and forced-convection produces the finest output with the highest dehydration rate and maximum quality among the different dryer categories. The dehydration rate is permanently increased by employing a dryer with a forced hot-convection capability. The performance examination for several types of solardrvers like 'direct, indirect, hybrid dryers and mixed mode' combined with the natural energy storage materials explored. The use of diverse types of sun dehydration technologies can greatly reduce postharvest damages for agro-products in underdeveloped nations. In addition, this paper summarizes the procedural guidelines for the development and



Fig. 11. The effective risks of the solar dryer systems implementation.

design of solar aided drying methods for agro-products. The usage of greenhouse dryers when it is integrated with hybrid solar-collectors is progressively recognized in the present research.

Novel thermal storage and desiccant materials are thus required for hybrid solar drying structures. But with the integration of NES materials with solar drying setups may increase the size of the solar dryer due to the low density of NES materials, which needs to be improved and more research is needed in this area. For food drying applications, the majority of the researchers considered water, sand, concrete and rocks. Pebbles, stones and rock are among the most suited materials, but highpressure drop and low thermal conductivity at high flow rates are important issues. Sand has the unusual ability to absorb and reflect heat quickly. As a result, when it was sunny, the temperature rose quickly. The stored heat is expected to be discharged quickly during gloomy weather or no-sunshine hours. With this, a successful solar energy system can be employed for pollution-free drying processes by focusing on these areas. Finally, the challenges, advantages and prospects of NES materials integrated with solar-drying applications are discussed. From an economic standpoint, the system may operate after the sun has set, allowing food to be dried constantly and at a lower cost. Future research could look into different pretreatment procedures to provide the chemical and physical changes needed to increase mass and heat transmission. In addition, research is needed to determine the best storage material to avoid overusing storage materials and reducing the setup's volume.

CRediT authorship contribution statement

Bade Venkata Suresh: Writing – original draft, Formal analysis, Data curation, Conceptualization. Yegireddi Shireesha: Writing – original draft, Formal analysis, Data curation. Teegala Srinivasa Kishore: Writing – original draft, Methodology, Investigation, Formal analysis, Conceptualization. Gaurav Dwivedi: Writing – review & editing, Visualization, Investigation, Data curation. Ali Torabi Haghighi: Writing – review & editing, Visualization, Investigation. Epari Ritesh Patro: Writing – review & editing, Visualization, Supervision, Project administration, Methodology, Conceptualization.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

No data was used for the research described in the article.

References

- M. R. Nukulwar, V. B. Tungikar, Recent development of the solar dryer integrated with thermal energy storage and auxiliary units, Therm. Sci. Eng. Prog. 29, 101192. https://doi.org/10.1016/j.tsep.2021.101192.
- [2] A.A. El-Sebaii, S.M. Shalaby, Solar drying of agricultural products: a review, Renew. Sustain. Energy Rev. 16 (2022) 37–43, https://doi.org/10.1016/jrser.2011.07.134, 2012.
- [3] Stockholm International Water Institute Sweden, Saving water: from field to fork: curbing losses and wastage in the food chain. https://hdl.handle.net/1 0535/5088, 2008. (Accessed 20 October 2022).
- [4] Food and Agriculture Organization Italy, Food wastage footprint: impacts on natural resources. https://www.fao.org/3/i3347e/i3347e.pdf, 2013. (Accessed 20 October 2022).
- [5] Waste & Resources Action Programme Uk, Food waste report 2: the food we waste. https://wrap.s3.amazonaws.com/the-food-we-waste-executive-summary. pdf, 2008. (Accessed 20 October 2022).
- [6] Table, What a waste! Surplus fresh foods research project. https://www.ta bledebates.org/research-library/what-waste-surplus-fresh-foods-research-projec t, 2007. (Accessed 20 October 2022).
- [7] Australian Institute of Family Studies, Australia, A snapshot of how Australian Families spend their time. https://aifs.gov.au/research/research-snapshots/sna

pshot-how-australian-families-spend-their-time, 2007. (Accessed 20 October 2022).

- [8] Voices Newsletter, Increasing post-harvest success for smallholder farmers. No: 79. http://scripts.farmradio.fm/wp-content/uploads/Voices_79.pdf, 2006 accessed on October 2022.
- [9] Food and Agriculture Organization Italy, The slate of the food insecurity in the world. https://www.fao.org/agrifood-economics/publications/detail/en/c/ 122054/, 2004 accessed on October 2022.
- [10] Stockholm International Water Institute Sweden, From field to fork: wastage of water in the food chain. https://siwi.org/wp-content/uploads/2015/09/pb_fr om_filed_to_fork_2008.pdf, 2007 accessed on October 2022.
 [11] Mohammad U.H. Joardder, Mahadi Hasan Masud, Food Preservation in
- Developing Countries: Challenges and Solutions, first ed., Springer, 2019. [12] Mohammad U.H. Joardder, Monjur Mourshed, Mahadi Hasan Masud,
- [122] Monaminato G.H. Joarduet, Monjur Monshed, Maladi Hasan Masud, Significance of bound water measurement, in: Mohammad U.H. Joardder, Monjur Mourshed, Mahadi Hasan Masud (Eds.), State of Bound Water: Measurement and Significance in Food Processing, Springer Cham, 2018, pp. 119–135.
- [13] M.S.B. Jahromi, V. Kalantar, H.S. Akhijahani, H. Kargarsharifabad, Recent progress on solar cabinet dryers for agricultural products equipped with energy storage using phase change materials, J. Energy Storage 51 (2022), 104434, https://doi.org/10.1016/j.est.2022.104434.
- [14] S. Singh, R.S. Gill, V.S. Hans, M. Singh, A novel active-mode indirect solar dryer for agricultural products: experimental evaluation and economic feasibility, Energy 222 (2021), 119956, https://doi.org/10.1016/j.energy.2021.119956.
- [15] N. Tippayawong, C. Tantakitti, S. Thavornun, Energy efficiency improvements in longan drying practice, Energy 33 (2008) 1137–1143, https://doi.org/10.1016/j. energy.2008.02.007.
- [16] J.C. Atuonwu, X. Jin, G.V. Stralen, H.C. D Antonious, J.B. Boxtel, Reducing energy consumption in food drying: opportunities in desiccant adsoprtion and other dehumidification stratergies, Procedia Food Sci 1 (2011) 1799–1805, https://doi.org/10.1016/j.profoo.2011.09.264.
- [17] P. Mehta, N. Bhatt, G. Bassan, Abd Elnaby Kabeel, Performance improvement and advancement studies of mixed-mode solar thermal dryers: a review, Environ. Sci. Pollut. Res. 29 (2022) 62822–62838, https://doi.org/10.1007/s11356-022-21736-3.
- [18] Abhay Menon, Valentina Stojceska, Savvas A. Tassou, A systematic review on the recent advances of the energy efficiency improvements in non-conventional food drying technologies, Trends Food Sci. Technol. 100 (2020) 67–76, https://doi. org/10.1016/j.tifs.2020.03.014.
- [19] O. Erkmen, T.F. Bozoglu, Food preservation by reducing water activity, in: O. Erkmen, T.F. Bozoglu (Eds.), Food Microbiology: Principles into Practice, John Wiley and Sons Ltd., 2016, pp. 44–58.
- [20] Selimefendigil Fatih, Ceylin Şirin, Hakan F. Öztop, Experimental performance analysis of a solar desalination system modified with natural dolomite powder integrated latent heat thermal storage unit, Sustainability 14 (2022) 2650, https://doi.org/10.3390/su14052650.
- [21] V.P. Chandramohan, P. Talukdar, Deformation of potato during convective drying, Appl. Mech. Mater. 592–594 (2014) 2728–2732. https://doi.org/10.402 8/www.scientific.net/AMM.592-594.2728.
- [22] A.R. Umayal Sundari, E. Veeramanipriya, Performance evaluation, morphological properties and drying kinetics of untreated Carica Papaya using solar hybrid dryer integrated with heat storage material, J. Energy Storage 55 (2022), 105679, https://doi.org/10.1016/j.est.2022.105679.
- [23] Samira Chouicha, Abdelghani Boubekri, Djamel Mennouche, Bouguetaia Hamza, Mohamed Hafed Berrbeuh, Samiha Bouhafs, Wahiba Rezzoug, Valorization study of treated deglet-nour dates by solar drying using three different solar driers, Procedia Energy 50 (2014) 907–916, https://doi.org/10.1016/j. egypro.2014.06.109.
- [24] M. Izadi, A. Hajjar, H.M. Alshehri, M. Sheremet, A.M. Galal, Charging process of a partially heated trapezoidal thermal energy storage filled by nano-enhanced PCM using controllable uniform magnetic field, Int. Commun. Heat Mass Tran. 138 (2022), 106349, https://doi.org/10.1016/j.icheatmasstransfer.2022.106349.
- [25] V.P. Chandramohan, P. Talukdar, Three dimensional numerical modeling of si-multaneous heat and moisture transfer in a moist object subjected to convective drying, Int. J. Heat Mass Tran. 53 (2010) 4638–4650, https://doi.org/ 10.1016/j.ijheatmasstransfer.2010.06.029.
- [26] M.C. Ndukwu, M. Ibeh, I. Ekop, U. Abada, P. Etim, L. Bennamoun, F. Abam, M. SimoTagne, A. Gupta, Analysis of the heat transfer coefficient, thermal effusivity and mathematical modelling of drying kinetics of a partitioned single pass low-cost solar drying of cocoyam chips with economic assessments, Energies 15 (2022) 4457, https://doi.org/10.3390/en15124457.
- [27] W. Muhlbauer, Present status of solar crop drying, Energy Agric. 5 (1986) 121–137, https://doi.org/10.1016/0167-5826(86)90013-6.
- [28] C.L. Hii, S.V. Jangam, S.P. Ong, A.S. Mujumdar, Solar Drying: Fundamentals, Applications and Innovations, first ed., TPR Group Publications, Singapore, 2012.
- [29] W.B. Chaouch, A. Khellaf, A. Mediani, M.E.A. Sliniani, A. Loumani, A. Harnid, Experimental investigation of an active direct and indirect solar dryer with sensible heat storage for camel meat drying in Saharan environment, Sol. Energy 174 (2018) 328–341, https://doi.org/10.1016/j.solener.2018.09.037.
- [30] Y. Yao, Y.X. Pang, S. Manickam, E. Lester, T. Wu, C.H. Pang, A review study on recent advances in solar drying: mechanisms, challenges and perspectives, Sol. Energy Mater. Sol. Cells 248 (2022), 111979, https://doi.org/10.1016/j. solmat.2022.111979.
- [31] V. Belessiotis, E. Delyannis, Solar drying, Sol. Energy 85 (2011) 1665–1691, https://doi.org/10.1016/j.solener.2009.10.001.

- [32] V. Tomar, G. Tiwari, B. Norton, Solar dryers for tropical food preservation: thermophysics of crops, systems and components, Sol. Energy 154 (2017) 2–13, https://doi.org/10.1016/j.solener.2017.05.066.
- [33] M.A. Basunia, T. Abe, Thin-layer solar drying characteristics of rough rice under natural convection, J. Food Eng. 47 (2001) 295–301, https://doi.org/10.1016/ S0260-8774(00)00133-3.
- [34] A.B. Lingayat, V.P. Chandraniohan, V.R.K. Raju, V. Meda, A review on indirect type solar dryers for agricultural crops - dryer setup, its performance, energy storage and important highlights, Appl. Energy 258 (2020), 114005, https://doi. org/10.1016/j.apenergy.2019.114005.
- [35] D. Jain, P. Tiwari, Performance of indirect through pass natural convective solar crop dryer with phase change thermal energy storage, Renew. Energy 80 (2015) 244–250, https://doi.org/10.1016/j.renene.2015.02.012.
- [36] A. El Khadraoui, S. Kooli, I. Hamdi, A. Farhat, Experimental investigation and economic evaluation of a new mixed mode solar greenhouse dryer for drying of red pepper and grape, Renew. Energy 77 (2017) 1–8, https://doi.org/10.1016/j. renene.2014.11.090.
- [37] M. Condori, G. Duraa, R. Echazu, R. Altobelli, Semi-industrial drying of vegetables using an array of large solar air collectors, Energy Sustain. Dev. 37 (2017) 1–9, https://doi.org/10.1016/j.esd.2016.11.004.
- [38] C.K.K. Sekyere, F.K. Forson, F.W. Adam, Experimental investigation of the drying characteristics of a mixed mode natural circulation solar crops dryer with back up heater, Renew. Energy 92 (2016) 532–542, https://doi.org/10.1016/j. renene.2016.02.020.
- [39] M.S.B. Jahromi, M. Iranmanesh, H.S. Akhijahani, Thermo-economic analysis of solar drying of Jerusalem artichoke (Helianthus tuberosus L.) integrated with evacuated tube solar collector and phase change material, J. Energy Storage 52 (2022), 104688, https://doi.org/10.1016/j.est.2022.104688.
- [40] B.K. Bala, J.L. Woods, Simulation of the indirect natural convection solar drying of rough rice, Sol. Energy 53 (1994) 259–266, https://doi.org/10.1016/0038-092X(94)90632-7.
- [41] R. Queiroz, A.L. Gabas, R.V.N. Telisl, Drying kinetics of tomato by using electric resistance and heat pump dryers, Dry. Technol. 22 (2004) 1603–1620, https:// doi.org/10.1081/DRT-200025614.
- [42] Capossio Juan Pablo, María Paula Fabani, Andrés Reyes-Urrutia, Rodrigo Torres-Sciancalepore, Yimin Deng, Baeyens Jan, Rosa Rodriguez, Germán Mazza, Sustainable solar drying of brewer's spent grains: a comparison with conventional electric convective drying, Processes 10 (2022) 339, https://doi.org/10.3390/ pr10020339.
- [43] A.G. Ferreira, A. Charbel, J.G. Silva, Experimental study of hybrid dryer, Therm. Eng. 6 (2007) 3–7, https://doi.org/10.5380/reterm.v6i2.61680.
- [44] B.M.A. Amer, K. Gottschalk, M.A. Hossain, Integrated hybrid solar drying system and its drying kinetics of chamomile, Renew. Energy 121 (2018) 539–547, https://doi.org/10.1016/j.renene.2018.01.055.
- [45] M.A. Eltawil, M.M. Azarn, A.O. Alghannam, Energy analysis of hybrid solar tunnel dryer with PV system and solar collector for drying mint (Mentha Viridis), J. Clean. Prod. 181 (2018), 352364, https://doi.org/10.1016/j. iclepro.2018.01.229.
- [46] D. Gudino Ayala, A. Calderon Topete, Pineapple drying using a new solar hybrid dryer, Energy Proc. 57 (2014) 1642–1650, https://doi.org/10.1016/j. egypro.2014.10.155.
- [47] S. Dhanushkodi, V.H. Wilson, K. Sudhakar, Mathematical modeling of drying behavior of cashew in a solar biomass hybrid dryer, Res. Eff. Technol. 3 (2017) 359–364, https://doi.org/10.1016/j.reffit.2016.12.002.
- [48] R.K. Aggarwal, S.S. Chandel, S. Gorjian, R. Chandel, Research outcome of sustainable solar drying technology dissemination for preserving perishable agriculture and horticulture crops in the North Western Himalayan region of India, Sustain, Energy Technol. Assess. 53 (2022), 102732, https://doi.org/ 10.1016/j.seta.2022.102732.
- [49] A. Reyes, A. Mahn, F. Cubillos, P. Huenulaf, Mushroom dehydration in a hybridsolar dryer, Energy Convers. Manag. 70 (2013) 31–39, https://doi.org/10.1016/j. enconman.2013.01.032.
- [50] C. Ramirez, M. Palacio, M. Carmona, Reduced model and comparative analysis of the thermal performance of indirect solar dryer with and without PCM, Energies 13 (2020) 5508, https://doi.org/10.3390/en13205508.
- [51] S. Mekhilef, S.Z. Faraniarzi, R. Saidur, Z. Salani, The application of solar technologies for sustainable development of agricultural sector, Renew. Sustain. Energy Rev. 18 (2013) 583–594, https://doi.org/10.1016/j.rser.2012.10.049.
- [52] S. Haldorai, S. Gurusaniy, M. Pradhapraj, A review on thermal energy storage systems in solar air heaters, Int. J. Energy Res. 43 (2019) 6061–6077, https://doi. org/10.1002/er.4379.
- [53] L.M. Bal, S. Satya, S.N. Naik, V. Meda, Review of solar dryers with latent heat storage systems for agricultural products, Renew. Sustain. Energy Rev. 15 (2011) 876–880, https://doi.org/10.1016/j.rser.2010.09.006.
- [54] K. Kant, A. Shukla, A. Sharnia, A. Kumar, A. Jain, Thermal energy storage based solar drying systems: a review, Innovat. Food Sci. Emerg. Technol. 34 (2016) 86–99, https://doi.org/10.1016/j.ifset.2016.01.007.
- [55] M. Esen, T. Ayhan, Development of a model compatible with solar assisted cylindrical energy storage tank and variation of stored energy with time for different phase change materials, Energy Convers. Manag. 37 (1996) 1775–1785, https://doi.org/10.1016/0196-8904(96)00035-0.
- [56] P.V. Alfiya, G.K. Rajesh, S. Murali, D.A. Delfiya, M.P. Samuel, M.V. Prince, Quality evaluation of solar and microwave dried shrimps – a comparative study on renewable and dielectric heating methods, Sol. Energy 246 (2022) 234–244, https://doi.org/10.1016/j.solener.2022.09.048.

- [57] S. Ayyappan, K. Mayilsamy, V.V. Sreenarayanan, Performance improvement studies in a solar greenhouse drier using sensible heat storage materials, Heat Mass Tran. 52 (2016) 459–467, https://doi.org/10.1007/s00231-015-1568-5.
- [58] T.F.N. Thoruwa, C.M. Johnstone, A.D. Grant, J.E. Smith, Novel, low cost CaCl2 based desiccants for solar crop drying applications, Renew. Energy 19 (2000) 513–520, https://doi.org/10.1016/S0960-1481(99)00072-5.
- [59] O. Badaoui, A. Djebli, S. Hanini, Solar drying of apple and orange waste: evaluation of a new thermodynamic approach, and characterization analysis, Renew. Energy 199 (2022) 1593–1605, https://doi.org/10.1016/j. renene.2022.09.098.
- [60] F. Ozgen, M. Esen, H. Esen, Experimental investigation of thermal performance of a double-flow solar air heater having aluminum cans, Renew. Energy 34 (2009) 2391–2398, https://doi.org/10.1016/j.renene.2009.03.029.
- [61] A.K. Bhardwaj, R. Kumar, R. Chauhan, Experimental investigation of the performance of a novel solar dryer for drying medicinal plants in Western Himalayan region, Sol. Energy 177 (2019) 395–407, https://doi.org/10.1016/j. solener.2018.11.007.
- [62] Halil Atalay, Performance analysis of a solar dryer integrated with the packed bed thermal energy storage (TES) system, Energy 172 (2019) 1037–1052, https://doi. org/10.1016/j.energy.2019.02.023.
- [63] S. Vijayan, T.V. Arjunan, A. Kuniar, Exergo-environmental analysis of an indirect forced convection solar dryer for drying bitter gourd slices, Renew. Energy 146 (2020) 2210–2223, https://doi.org/10.1016/j.renene.2019.08.066.
- [64] W. Amjad, M.A. Raza, F. Asghar, A. Munir, F. Mahmood, S.N. Husnain, M. I. Hussain, J.T. Kim, Advanced exergy analyses of a solar hybrid food dehydrator, Energies 15 (2022) 1505, https://doi.org/10.3390/en15041505.
- [65] M.R. Patel, N.L. Panwar, Drying kinetics, quality assessment and socio environmental evaluation of solar dried underutilized arid vegetable Cucumis callosus, Energy Nexus 7 (2022), 100128, https://doi.org/10.1016/j. nexus.2022.100128.
- [66] E. Baniasadi, S. Ranjbar, O. Boostanipour, Experimental investigation of the performance of a mixed-mode solar dryer with thermal energy storage, Renew. Energy 112 (2017) 143–150, https://doi.org/10.1016/j.renene.2017.05.043.
- [67] D.V.N. Lakshmi, P. Muthukumar, A. Layek, P.K. Nayak, Drying kinetics and quality analysis of black turmeric (Curcuma caesia) drying in a mixed mode forced convection solar dryer integrated with thermal energy storage, Renew. Energy 120 (2018) 23–34, https://doi.org/10.1016/j.renene.2017.12.053.
- [68] S. Shyam, Sablani, Drying of fruits and vegetables: retention of nutritional/ functional quality, Dry, Technol. 24 (2006) 123–135, https://doi.org/10.1080/ 07373930600558904.
- [69] L.M. Raghavi, J.A. Moses, C. Anandharaniakrishnan, Refractance window drying of foods: a review, J. Food Eng. 222 (2018) 267–275, https://doi.org/10.1016/j. jfoodeng.2017.11.032.
- [70] G.W. Lim, S. Jafarzadeh, M.R. Norazatul Hanini, Kinetic study, optimization and comparison of sun drying and superheated steam drying of asam gelugor (Garcinia cambogia), Food Res. 4 (2020) 396–406, https://doi.org/10.26656/ fr.2017.4(2).288.
- [71] S.V. Jangam, A.S. Mujumdar, Basic concepts and definitions, in: S.V. Jangam, C. L. Law, A.S. Mujumdar (Eds.), Drying of Foods, Vegetables and Fruits, NUS, Singapore, 2010, pp. 1–30.
- [72] T. Zimmer, A. Rudi, S. Glöser Chahoud, F. Schultmann, Techno-economic analysis of intermediate pyrolysis with solar drying: a Chilean case study, Energies 15 (2022) 2272, https://doi.org/10.3390/en15062272.
- [73] M.A. Leon, S. Kumar, S.C. Bhattacharya, A comprehensive procedure for performance evaluation of solar food dryers, Renew. Sustain. Energy Rev. 6 (2002) 367–393, https://doi.org/10.1016/S1364-0321(02)00005-9.
- [74] A. Vega, E. Unbe, R. Lemus, M. Miranda, Hot-air drying characteristics of Aloe vera (Aloe barbadensis Miller) and influence of temperature on kinetic parameters, LWT-Food Sci. Technol. 40 (2007) 1698–1707, https://doi.org/ 10.1016/j.lwt.2007.01.001.
- [75] A.B. Lingayat, V.P. Chandramohan, V.R.K. Raju, A. Kuniar A, Development of indirect type solar dryer and experiments for estimation of drying parameters of apple and watermelon, Therm. Sci. Eng. Prog. 16 (2020), 100477, https://doi. org/10.1016/j.tsep.2020.100477.
- [76] S. Sharma, K. Dhalsamant, P.P. Tripatliy, R.K. Manepally, Quality analysis and drying characteristics of turmeric (Curcuma longa L.) dried by hot air and direct solar dryers, Lebensm. Wiss. Technol. 138 (2021) 1106S7, https://doi.org/ 10.1016/j.lwt.2020.110687.
- [77] H.A. Leniger, W.A. Beverloo, Food Process Engineering, first ed., D. Reidel Publishing Company, Dordrecht, Holland, 1975.
- [78] Y.M. Gallali, Y.S. Abujnah, F.K. Bannani, Preservation of fruits and vegetables using solar dryer: a comparative study of natural and solar drying. III; Chemical analysis and sensory evaluation data of the dried samples (grapes, figs, tomatoes and onions), Renew. Energy 19 (1999) 203–212, https://doi.org/10.1016/S0960-1481(99)00032-4.
- [79] S. Ranganna, Handbook of Analysis and Quality Control for Fruits and Vegetable Products, first ed., Tata McGraw-Hill Publishing Company, New Delhi, 1986.
- [80] A. R. Oliveira Fernanda, L. Ilincanu, Rehydration of dried plant tissues: basic concepts and mathematical model–ling, in: A. R. Oliveira Fernanda, Jorge C. Oliveira (Eds.), Processing Foods: Quality Optimization and Process Assessment, CRC Press, 1999.
- [81] P.P. Lewicki, Some remarks on rehydration of dried foods, J. Food Eng. 36 (1998) 81–87, https://doi.org/10.1016/S0260-8774(98)00022-3.
- [82] D.M. Khedkar, S.K. Roy, Histological evidence for the reconstitutional property of dried/dehydrated raw mango slices, J. Food Sci. Technol. 17 (1990) 276–277.

- [83] H. Karami, M. Kaveh, I. Golpour, E. Khalife, R. Rusinek, B. Dobrzański Jr., M. Gancarz, Thermodynamic evaluation of the forced convective hybrid-solar dryer during drying process of rosemary (rosmarinus officinalis L.) leaves, Energies 14 (2021) 5835, https://doi.org/10.3390/en14185835.
- [84] S. Toure, S. Kibangu Nkembo, Comparative study of natural solar drying of cassava, banana and mango, Renew. Energy 29 (2004) 975–990, https://doi.org/ 10.1016/j.renene.2003.09.013.
- [85] C.L. Hii, R. Abdul Rahman, S. Jinap, Y.B. Che Man, Quality of cocoa beans dried using a direct solar dryer at different loadings, J. Sci. Food Agric. 86 (2006) 1237–1243, https://doi.org/10.1002/jsfa.2475.
- [86] Y. Li, H.F. Li, Y.J. Dai, S.F. Gao, L. Wei, Z.L. Li, I.G. Odinez, R.Z. Wang, Experimental investigation on a solar assisted heat pump in-store drying system, Appl. Therm. Eng. 31 (2011) 1718–1724, https://doi.org/10.1016/j. applthermaleng.2011.02.014.
- [87] M.S. Sodha, D. Aman, P.K. Bansal, S.B. Sharman, An analytical and experimental study of open sun drying and a cabinet tyre drier, Energy Convers. Manag. 25 (1985) 263–271, https://doi.org/10.1016/0196-8904(85)90042-1.
- [88] P.N. Sarsavadia, Development of a solar-assisted dryer and evaluation of energy requirement for the drying of onion, Renew. Energy 32 (2007) 2529–2547, https://doi.org/10.1016/j.renene.2006.12.019.
- [89] K. Demir, K. Sacilik, Solar drying of Ayas tomato using a natural convection solar tunnel dryer, J. Food Agric. Environ. 8 (2010) 7–12.
- [90] B.K. Bala, M.R.A. Mondol, B.K. Biswas, B.L. Das Chowdury, S. Janjai, Solar drying of pineapple using solar tunnel drier, Renew. Energy 28 (2003) 183–190, https:// doi.org/10.1016/S0960-1481(02)00034-4.
- [91] K. Lutz, W. Mühlbauer, J. Müller, G. Reisinger, Development of a multi-purpose solar crop dryer for arid zones, Sol. Wind Technol. 4 (4) (1987) 417–424, https:// doi.org/10.1016/0741-983X(87)90016-6.
- [92] A. Fudholi, K. Sopian, M.H. Ruslan, M.A. Alghoul, M.Y. Sulaiman, Review of solar dryers for agricultural and marine products, Renew. Sustain. Energy Rev. 14 (2010) 1–30, https://doi.org/10.1016/j.rser.2009.07.032.
- [93] A.K. Elicin, K. Sacllik, An experimental study for solar tunnel drying of apple, Tarini Bilini Derg 11 (2005) 207–211, https://doi.org/10.1501/TARIMBIL_0000000421.
- [94] K.P. Thanvi, P.C. Pande, Development of a low-cost solar agricultural dryer for arid regions of India, Energy Agric. 6 (1987) 35–40, https://doi.org/10.1016/ 0167-5826(87)90020-9.
- [95] P.A. Potdukhe, S.B. Thonibre, Development of a new type of solar dryer its mathematical modelling and experimental evaluation, Int. J. Energy Res. 32 (2008) 765–782, https://doi.org/10.1002/er.1387.
- [96] A. El-Beltagy, G.R. Gamea, A.H.A. Essa, Solar drying characteristics of strawberry, J. Food Eng. 78 (2007) 456–464, https://doi.org/10.1016/j. ifoodeng.2005.10.015.
- [97] A. Abene, V. Dubois, M. Le Ray, A. Ouagued, Study of a solar air flat place collector: use of obstacles and application for the drying of grape, J. Food Eng. 65 (2004) 15–22, https://doi.org/10.1016/j.jfoodeng.2003.11.002.
- [98] L. Blanco Cano, A. Soria Verdugo, L.M. Garcia Gutierrez, U. Ruiz Rivas, Modeling the thin-layer drying process of Granny Smith apples: application in an indirect solar dryer, Appl. Therm. Eng. 108 (2016) 1086–1094, https://doi.org/10.1016/ j.applthermaleng.2016.08.001.
- [99] K.E.J. Al-Juaniily, A.J.N. Khalifa, T.A. Yassen, Testing of the performance of a fruit and vegetable solar drying system in Iraq, Desalination 209 (2007) 163–170, https://doi.org/10.1016/j.desal.2007.04.026.
- [100] A. Djebli, S. Hanini, O. Badaoui, B. Haddad, A. Benhamou, Modeling and comparative analysis of solar drying behavior of potatoes, Renew. Energy 145 (2020) 1494–1506, https://doi.org/10.1016/j.renene.2019.07.083.
 [101] A. Djebli, S. Hanim, O. Badaoui, M. Boumahcu, A new approach to the
- [101] A. Djebli, S. Hanim, O. Badaoui, M. Boumahcu, A new approach to the thermodynamics study of drying tomatoes in mixed solar dryer, Sol. Energy 193 (2019) 164–174, https://doi.org/10.1016/j.solener.2019.09.057.
- [102] Z.A.F. Mahdizadeh, A. Zomorodian, A study of the effect of solar drying system on rice quality, J. Agric. Sci. Technol. 11 (2009) 527–534, 20.1001.1.16807073.2009.11.5.10.7.
- [103] L.F. Hidalgo, M.N. Candido, K. Nishioka, J.T. Freire, G.N.A. Vieira, Natural and forced air convection operation in a direct solar dryer assisted by photovoltaic module for drying of green onion, Sol. Energy 220 (2021) 24–34, https://doi.org/ 10.1016/j.solener.2021.02.06.
- [104] A. Madhlopa, G. Ngwalo, Solar dryer with thermal storage and biomass-backup heater, Sol. Energy 81 (2007) 449–462, https://doi.org/10.1016/j. solener.2006.08.008.
- [105] M.H. Ruslan, M.Y. Othman, Baharuddin Yatim, Kamaruzzaman Sopian, M.I. Ali, Z. Ibarahim, Photovoltaic assisted solar drying system, in: Proc. Int. Symp. Renew. Energy: Environ. Protection and Energy Solut., Malaysia, 2006, pp. 681–687.
- [106] A. Tiwari, A review on solar drying of agricultural produce, J. Food Process. Technol. 7 (2016) 1–12, https://doi.org/10.4172/2157-7110.1000623.
- [107] M.A. Aravindh, A. Sreekumar, Solar drying-A sustainable way of food processing, in: A.A. Sharma, S.K. Kar (Eds.), Energy Sustain. Through Gr. Energy, Springer, New Delhi, 2014, pp. 27–46, https://doi.org/10.1007/978-81-322-2337-5_2.
- [108] S. Kumar, Assessment of renewables for energy security and carbon mitigation in Southeast Asia: the case of Indonesia and Thailand, Appl. Energy 163 (2016) 63–70, https://doi.org/10.1016/j.apenergy.2015.11.019.
- [109] P. Purohit, A. Kumar, T.C. Kandpal, Solar drying vs. open sun drying: a framework for financial evaluation, Sol. Energy 80 (2006) 1568–1579, https://doi.org/ 10.1016/j.solener.2005.12.009.
- [110] J. Kaewkiew, S. Nabnean, S. Janjai, Experimental investigation of the performance of a large-scale greenhouse type solar dryer for drying chili in

Thailand, Process Eng. 32 (2012) 433–439, https://doi.org/10.1016/j. proeng.2012.01.1290.

- [111] E.A. Abdelaziz, R. Saidur, S. Mekhilef, A review on energy saving strategies in industrial sector, Renew. Sustain. Energy Rev. 15 (2011) 150–168, https://doi. org/10.1016/j.rser.2010.09.003.
- [112] G. Pirasteh, S. Rahman, S. Rahman, N.A. Rahim, A review on development of solar drying applications, Renew. Sustain. Energy Rev. 31 (2014) 133–148, https://doi.org/10.1016/j.rser.2013.11.052.
- [113] S. Carley, State renewable energy electricity policies: an empirical evaluation of effectiveness, Energy Pol. 37 (2009) 3071–3081, https://doi.org/10.1016/j. enpol.2009.03.062.
- [114] Paris: REN21 Secretariat, REN21. 2017, renewables 2017 global status report. htt ps://www.ren21.net/wp-content/uploads/2019/05/GSR2017_Full-Report_Engl ish.pdf, 2017. (Accessed 20 October 2022).
- [115] WHO Regional Office Europe, Health effects of transport-related air pollution. https://www.euro.who.int/_data/assets/pdf_file/0006/74715/E86650.pdf, 2005. (Accessed 20 October 2022).
- [116] M.M. Morad, M.A. El-Shazly, K.I. Wasfy, Hend AM. El-Maghawry, Thermal analysis and performance evaluation of a solar tunnel greenhouse dryer for drying peppermint plants, Renew. Energy 101 (2017) 992–1004, https://doi.org/ 10.1016/j.renene.2016.09.042.
- [117] A.R. Eswara, M. Ramakrishnarao, Solar energy in food processing a critical appraisal, J. Food Sci. Technol. 50 (2013) 209–227, https://doi.org/10.1007/ s13197-012-0739-3.
- [118] A. Arata, V.K. Sharma, G. Spagna, Performance evaluation of solar assisted dryers for low temperature drying application—II, Experimental results, Energy Convers. Manag. 34 (1993) 417–426, https://doi.org/10.1016/0196-8904(93) 90091-N.
- [119] M.L. Liu, S.Y. Wang, K.R. Li, Study of the solar energy drying device and its application in traditional Chinese medicine in drying, Int. J. Clin. Med. 6 (2015) 271–280, https://doi.org/10.4236/ijcm.2015.64034.
- [120] R.D. Piacentini, A.S. Mujumdar, Climate change and the drying of agricultural products, Dry. Technol. 27 (2009) 629–635, https://doi.org/10.1080/ 07373930902820770.
- [121] O.V. Ekechukwu, B. Norton, Review of solar-energy drying systems II: an overview of solar drying technology, Energy Convers. Manag. 40 (1999) 615–655, https://doi.org/10.1016/S0196-8904(98)00093-4.
- [122] S. Vijayan, T.V. Arjunan, A. Kumar, M.M. Matlieswaran, Experimental and thermal performance investigations on sensible storage based solar air heater, J. Energy Storage 31 (2020), 101620, https://doi.org/10.1016/j. est.2020.101620.
- [123] A.B. Lingayat, R. Zachariah, A. Modi, Current status and prospect of integrating solar air heating systems for drying in various sectors and industries, Sustain. Energy Technol. Assessments 52 (2022), 102274, https://doi.org/10.1016/j. seta.2022.102274.
- [124] P. Tatsidjodoung, N. Le Pierres, L. Luo, A review of potential materials for thermal energy storage in building applications, Renew. Sustain. Energy Rev. 18 (2013) 327–349, https://doi.org/10.1016/j.rser.2012.10.025.
- [125] V.R. Mugi, V.P. Chandramohan, Shrinkage, effective diffusion coefficient, surface transfer coefficients and their factors during solar drying of food products - a review, Sol. Energy 229 (2021) 84–101, https://doi.org/10.1016/j. solener.2021.07.042.
- [126] H. Atalay, Assessment of energy and cost analysis of packed bed and phase change material thermal energy storage systems for the solar energy-assisted drying process, Sol. Energy 198 (2020) 124–138, https://doi.org/10.1016/j. solener.2020.01.05.
- [127] S. Vijayan, T.V. Arjunan, Anil Kumar, Mathematical modeling and performance analysis of thin layer drying of bitter gourd in sensible storage based indirect solar dryer, Innovat. Food Sci. Emerg. Technol. 36 (2016) 59–67, https://doi.org/ 10.1016/j.ifset.2016.05.014.
- [128] M. Mohanraj, P. Chandrasekar, Performance of a solar drier with and without heat storage material for copra drying, Int. J. Global Energy Issues 31 (2009) 112–121, https://doi.org/10.1504/IJGEI.2009.023888.
- [129] S. Aboul-Enein, A.A. El-Sebaii, M.R.I. Ramadan, H.G. El-Gohary, Parametric study of a solar air heater with and without diermai storage for solar drying applications, Renew. Energy 21 (2000) 505–522, https://doi.org/10.1016/ S0960-148K00) 00092-6.
- [130] H. Wang, M. Torki, H.W. Xiao, V. Orsat, G.S.V. Raghavan, Z.L. Liu, W.J. Peng, X. L. Fang, Multi-objective analysis of evacuated tube solar-electric hybrid drying setup for drying lotus bee pollen, Renew. Sustain. Energy Rev. 168 (2022), 112822, https://doi.org/10.1016/j.rser.2022.112822.
- [131] Y.X. Bi, S. Zielinska, J.B. Ni, X.X. Li, X.F. Xue, W.L. Tian, W.J. Peng, X.M. Fang, Effects of hot-air drying temperature on drying characteristics and color deterioration of rape bee pollen, Food Chem. 16 (2022), 100464, https://doi.org/ 10.1016/j.fochx.2022.100464.
- [132] I. Dincer, Thermal energy storage systems as a key technology in energy conservation, Int. J. Energy Res. 26 (2002) 567–588, https://doi.org/10.1002/ er.805.
- [133] I. Dincer, M.A. Rosen, Thermal Energy Storage: Systems and Applications, second ed., John Wiley & Sons Ltd., 2011.
- [134] L.M. Bal, S. Satya, S.N. Naik, S. N, Solar dryer with thermal energy storage systems for drying agricultural food products: a review, Renew. Sustain. Energy Rev. 14 (2010) 2298–2314, https://doi.org/10.1016/j.rser.2010.04.014.
- [135] N.M. Ortiz Rodríguez, M. Condorí, G. Durán, O. García Valladares, Solar Drying Technologies: a review and future research directions with a focus on

agroindustrial applications in medium and large scale, Appl. Therm. Eng. 215 (2022), 118993, https://doi.org/10.1016/j.applthermaleng.2022.118993.

- [136] M. Mohanraj, P. Chandrasekar, Drying of copra in a forced convection solar drier, Biosyst. Eng. 99 (2003) 604–607, https://doi.org/10.1016/j. biosystemseng.2007.12.004.
- [137] P.M. Chauhan, C. Choudhury, H.P. Garg, Comparative performance of coriander dryer coupled to solar air heater and solar air-heater-cum-rockbed storage, Appl. Therm. Eng. 16 (1996) 475–486, https://doi.org/10.1016/1359-4311(95)00038-
- [138] H. El Ferouali, A. Zoukit, I. Salhi, S. Doubabi, N. Abdenouri, Investigation of the efficiency of a solar dryer with thermal storage by rock salt and by pebble bed, Int. J. Renew. Energy Technol. 10 (2019) 283–300, https://doi.org/10.1504/ ijret.2019.102845.
- [139] Y. Tian, C.Y. Zhao, A review of solar collectors and thermal energy storage in solar thermal applications, Appl. Energy 104 (2013) 533–553, https://doi.org/ 10.1016/j.apenergy.2012.11.051.
- [140] N. Philip, S. Duraipandi, A. Sreekumar, Techno-economic analysis of greenhouse solar dryer for drying agricultural produce, Renew. Energy 199 (2022) 613–627, https://doi.org/10.1016/j.renene.2022.08.148.
- [141] A.A. El-Sebaii, S. Aboul-Enein, M.R.I. Ramadan, H.G. El-Gohary, Experimental investigation of an indirect type natural convection solar dryer, Energy Convers. Manag. 43 (2002) 2251–2266, https://doi.org/10.1016/S0196-8904(01)00152-2
- [142] S. Ayyappan, K. Mayilsamy, V.V. Sreenarayanan, Perfomiance improvement studies in a solar greenhouse drier using sensible heat storage materials, Heat Mass Tran. 52 (2016) 459–467, https://doi.org/10.1007/s00231-015-1568-5.
- [143] S. Boughali, H. Benmoussa, B. Bouchekima, D. Mennouche, H. Bouguettaia, D. Bechki, Crop drying by indirect active hybrid solar–Electrical dryer in the eastern Algerian Septentrional Sahara, Sol. Energy 83 (2009) 2223–2232, https:// doi.org/10.1016/j.solener.2009.09.006.
- [144] M. Mohanraj, P. Chandrasekar, Performance of a forced convection solar drier integrated with gravel as heat storage material for chili drying, J. Eng. Sci. Technol. 4 (2009) 305–314.
- [145] S.K. Sansaniwal, V. Sharma, J. Mathur, Energy and exergy analyses of various typical solar energy applications: a comprehensive review, Renew. Sustain. Energy Rev. 82 (2018) 1576–1601, https://doi.org/10.1016/j.rser.2017.07.003.
- [146] Ankit Srivastava, Abhishek Anand, Amritanshu Shukla, Anil Kumar, D. Buddhi, Atul Sharma, A comprehensive overview on solar grapes drying: modeling, energy, environmental and economic analysis, Sustain, Energy Technol. Assess. 47 (2021), 101513, https://doi.org/10.1016/j.seta.2021.101513.
- [147] Ravinder Kumar, A critical review on energy, exergy, exergoeconomic and economic (4-E) analysis of thermal power plants, Int. J. Eng. Sci. Technol. 20 (2017) 283–292, https://doi.org/10.1016/j.jestch.2016.08.018.
- [148] Natural Resources, Scientific research, irvine, USA, exergetic analysis of solar energy drying systems. http://repository.fuoye.edu.ng/handle/123456789/1232, 2011.
- [149] A. Fudholi, R. Yendra, D.F. Basri, M.H. Ruslan, K. Sopian, Energy and exergy analysis of hybrid solar drying system, Contemporary Eng. Sci. 9 (2016) 215–223, https://doi.org/10.12988/ces.2016.512323.
- [150] Y. Baradey, M.N.A. Hawlader, A.F. Ismail, M. Hrairi, M.I. Rapi, Solar drying of fruits and vegetables, Int. J. Recent Dev. Eng. Technol. 5 (2016) 1–6.
- [151] M. Yahya, A. Fudlioli, K. Sopian, Energy and exergy analyses of solar-assisted fluidized bed drying integrated with biomass furnace, Renew. Energy 105 (2017) 22–29, https://doi.org/10.1016/j.renene.2016.12.049.
- [152] Kumar Mahesh, Sunil Kumar Sansaniwal, Pankaj Khatak, Progress in solar dryers for drying various commodities, Renew. Sustain. Energy Rev. 55 (2016) 346–360, https://doi.org/10.1016/j.rser.2015.10.158.
- [153] Abhay Lingayat, V.P. Chandramohan, V.R.K. Raju, Design, development and performance of indirect type solar dryer for banana drying, Energy Proc. 109 (2017) 409–416, https://doi.org/10.1016/j.egypro.2017.03.041.
- [154] S. Abubakar, S. Umaru, M.U. Kaisan, U.A. Umar, B. Ashok, K. Nanthagopal, Development and performance comparison of mixed-mode solar crop dryers with and without thermal storage, Renew. Energy 128 (2018) 285–298, https://doi. org/10.1016/j.renene.2018.05.049.
- [155] Om Prakash, Anil Kumar, Vinod Laguri, Performance of modified greenhouse dryer with thermal energy storage, Energy Rep. 2 (2016) 155–162, https://doi. org/10.1016/j.egyr.2016.06.003.
- [156] Traiphop Phahom, Singhanat Phoungchandang, William L. Kerr, Effects of steammicrowave blanching and different drying processes on drying characteristics and quality attributes of Thunbergia laurifolia Linn. Leaves, J. Sci. Food Agric. 97 (2017) 3211–3219, https://doi.org/10.1002/jsfa.8167.
- [157] I. Daniel Onwude, Norhashila Hashim, Rimfiel B. Janius, Nazmi Mat Nawi, Khalina Abdan, Modeling the thin-layer drying of fruits and vegetables: a review, Compr. Rev. Food Sci. Food Saf. 15 (2016) 599–618, https://doi.org/10.1111/ 1541-4337.12196.
- [158] Aysegul Abusoglu, Mehmet Kanoglu, Exergoeconomic analysis and optimization of combined heat and power production: a review, Renew. Sustain. Energy Rev. 13 (2009) 2295–2308, https://doi.org/10.1016/j.rser.2009.05.004.
- [159] Adarsh Abi Mathew, T. Venugopal, Solar power drying system: a comprehensive assessment on types, trends, performance and economic evaluation, Int. J. Ambient Energy 42 (2021) 96–119, https://doi.org/10.1080/ 01430750.2018.1507933.
- [160] Aymen Elkhadraoui, Sami Kooli, Ilhem Hamdi, Abdelhamid Farhat, Experimental investigation and economic evaluation of a new mixed-mode solar greenhouse dryer for drying of red pepper and grape, Renew. Energy 77 (2015) 1–8, https:// doi.org/10.1016/j.renene.2014.11.090.

- [161] El Hage, Hicham, Herez Amal, Mohamad Ramadan, Bazzi Hassan, Mahmoud Khaled, An investigation on solar drying: a review with economic and environmental assessment, Energy 157 (2018) 815–829, https://doi.org/ 10.1016/j.energy.2018.05.197.
- [162] P. Purohit, T.C. Kandpal, Solar crop dryer for saving commercial fuels: a technoeconomic evaluation, Int. J. Ambient Energy 26 (2005) 3–12, https://doi.org/ 10.1080/01430750.2005.9674966.
- [163] Mustapha, Moshood Keke, Abdulbashir Femi Salako, Sunmola Kayode Ademola, Ifeoluwa Abimbola Adefila, Qualitative performance and economic analysis of low cost solar fish driers in Sub-Saharan Africa, J. Fisheries. 2 (2014) 64–69, https://doi.org/10.17017/j.fish.61.
- [164] Guruprasad Alva, Yaxue Lin, Guiyin Fang, An overview of thermal energy storage systems, Energy 144 (2018) 341–378, https://doi.org/10.1016/j. energy.2017.12.037.
- [165] Antoni Gil, Marc Medrano, Ingrid Martorell, Ana Lázaro, Pablo Dolado, Belén Zalba, F. Luisa, Cabeza, State of the art on high temperature thermal energy storage for power generation, Part 1—concepts, materials and modellization, Renew. Sustain. Energy Rev. 14 (2010) 31–55, https://doi.org/10.1016/j. rser.2009.07.035.
- [166] A. Kylili Angeliki, Pari fokaides, andreas ioannides, soteris kalogirou, environmental assessment of solar thermal systems for the industrial sector, J. Clean. Prod. 176 (2018) 99–109, https://doi.org/10.1016/j. jclepro.2017.12.150.
- [167] C. Catorze, A.P. Tavares, P. Cardão, A. Castro, M.E. Silva, D.W. Ferreira, I. Brás, Study of a solar energy drying system—energy savings and effect in dried food quality, Energy Rep. 8 (2022) 392–398, https://doi.org/10.1016/j. evyr.2022.01.070.
- [168] A. Reyes, E. Gatica, L. Henríquez-Vargas, N. Pailahueque, Modeling of sawdust drying in spouted beds using solar energy and phase change materials, J. Energy Storage 51 (2022), 104441, https://doi.org/10.1016/j.est.2022.104441.
- [169] P. Singh, M.K. Gaur, A review on role of solar drying technology in sustainable development, in: M. Pandit, L. Srivastava, R. Venkata Rao, J. Bansal (Eds.), Intelligent Computing Applications for Sustainable Real-World Systems, Proceedings in Adaptation, Learning and Optimization, Springer Cham, 2020, pp. 18–27, https://doi.org/10.1007/978-3-030-44758-8_3.
- [170] S. Murali, P.R. Amulya, P.V. Alfiya, D.A. Delfiya, M.P. Samuel, Design and performance evaluation of solar-LPG hybrid dryer for drying of shrimps, Renew. Energy 147 (2020) 2417–2428, https://doi.org/10.1016/j.renene.2019.10.002.
- [171] E. Arslan, M. Aktas, 4E analysis of infrared-convective dryer powered solar photovoltaic thermal collector, Sol. Energy 208 (2020) 46–57, https://doi.org/ 10.1016/j. solener.2020.07.071.
- [172] R. Daghigh, R. Shahidian, H. Oramipoor, A multistate investigation of a solar dryer coupled with photovoltaic thermal collector and evacuated tube collector, Sol. Energy 199 (2020) 694–703, https://doi.org/10.1016/j. solener.2020.02.069.
- [173] H. Chen, S. Qi, X. Tan, The improvement pathway for industrial energy efficiency under sustainability perspective, Sustain. Energy Technol. Assessments 51 (2022), 101949, https://doi.org/10.1016/j.seta.2022.101949.
- [174] M. Iranmanesh, H. Samimi Akhijahani, M.S. Barghi Jahromi, CFD modeling and evaluation the performance of a solar cabinet dryer equipped with evacuated tube solar collector and thermal storage system, Renew. Energy 145 (2020) 1192–1213, https://doi.org/10.1016/j.renene.2019.06.038.
- [175] D. Kong, Y. Wang, M. Li, V. Keovisar, M. Huang, Q. Yu, Experimental study of solar photovoltaic/thermal (PV/T) air collector drying performance, Sol. Energy 208 (2020) 978–989, https://doi.org/10.1016/j.solener.2020.08.067.
- [176] B. Chauhan Yagnesh, P.P. Rathod, A comprehensive review of the solar dryer, Int. J. Ambient Energy 41 (2020) 348–367, https://doi.org/10.1080/ 01430750.2018.1456960.
- [177] Ssemwanga Mohammed, Nakiguli Fatumah, Nasejje Shadia, Drying performance and economic analysis of novel hybrid passive-mode and active-mode solar dryers for drying fruits in East Africa, J. Stored Prod. Res. 88 (2020), 101634, https:// doi.org/10.1016/j.jspr.2020.101634.
- [178] A. Singh, J. Sarkar, R.R. Sahoo, Experimental performance analysis of novel indirect-expansion solar-infrared assisted heat pump dryer for agricultural products, Sol. Energy 206 (2020) 907–917, https://doi.org/10.1016/j. solener.2020.06.065.
- [179] Patchimaporn Udomkun, Sebastian Romuli, Steffen Schock, Busarakorn Mahayothee, Murat Sartas, Tesfamicheal Wossen, Emmanuel Njukwe, Bernard Vanlauwe, Joachim Müller, Review of solar dryers for agricultural products in Asia and Africa: an innovation landscape approach, J. Environ. Manag. 268 (2020), 110730, https://doi.org/10.1016/j.jenvman.2020.110730.
- [180] Z. Deng, M. Li, T. Xing, J. Zhang, Y. Wang, Y. Zhang, A literature research on the drying quality of agricultural products with using solar drying technologies, Sol. Energy 229 (2021) 69–83, https://doi.org/10.1016/j.solener.2021.07.041.
- [181] L.C. Hawa, U. Ubaidillah, S.A. Mardiyani, A.N. Laily, N.I. W Yosika, F.N. Afifah, Drying kinetics of cabya (Piper retrofractum Vahl) fruit as affected by hot water blanching under indirect forced convection solar dryer, Sol. Energy 214 (2021) 588–598, https://doi.org/10.1016/j.solener.2020.12.004.
- [182] N.M. Chinenye, D.I. Onyenwigwe, F. Abam, B. Lamrani, M. Simo-Tagne, N. Bekkioui, Z. Said, Influence of hot water blanching and saline immersion period on the thermal effusivity and the drying kinetics of hybrid solar drying of sweet potato chips, Sol. Energy 240 (2022) 176–192, https://doi.org/10.1016/j. solener.2022.05.026.
- [183] E. Chao, J. Li, L. Fan, L, Influence of combined freeze-drying and far-infrared drying technologies on physicochemical properties of seed-used pumpkin, Food Chem. 398 (2023), 133849, https://doi.org/10.1016/j.foodchem.2022.133849.

- [184] H.H. Al-Kayiem, A.A. Gitan, Flow uniformity assessment in a multi-chamber cabinet of a hybrid solar dryer, Sol. Energy 224 (2021) 823–832, https://doi.org/ 10.1016/j.solener.2021.06.058.
- [185] L.A. Gomes, R.F. Gonçalves, M.F. Martins, C.N. Sogari, Assessing the suitability of solar dryers applied to wastewater plants: a review, J. Environ. Manag. 326 (2023), 116640, https://doi.org/10.1016/j.jenvman.2022.116640.
- [186] A. Heydari, Experimental analysis of hybrid dryer combined with spiral solar air heater and auxiliary heating system: energy, exergy and economic analysis, Renew. Energy 198 (2022) 1162–1175, https://doi.org/10.1016/j. renene.2022.08.110.
- [187] H.S. Dheyab, M.S. Al-Jethelah, M. Baccar, Closed collector-storage solar air heating system thermal performance: an experimental study, Energy Sources, Part A Recovery, Util. Environ. Eff. 44 (2022) 10006–10023, https://doi.org/ 10.1080/15567036.2022.2144965.
- [188] R.E. Predolin, M.C. Ito, G.L. Palma, V.L. Scalon, An experimental and numerical investigation of absorber positioning in a natural convection solar drying system, Sol. Energy 243 (2022) 431–442, https://doi.org/10.1016/j. solener.2022.08.012.
- [189] P. Singh, M.K. Gaur, Heat transfer analysis of hybrid active greenhouse solar dryer attached with evacuated tube solar collector, Sol. Energy 224 (2021) 1178–1192, https://doi.org/10.1016/j.solener.2021.06.050.
- [190] M. Simo-Tagne, H.D.T. Etala, A.T. Tagne, M.C. Ndukwu, M. El Marouani, Energy, environmental and economic analyses of an indirect cocoa bean solar dryer: a comparison between natural and forced convections, Renew. Energy 187 (2022) 1154–1172, https://doi.org/10.1016/j.renene.2022.02.015.
- [191] A.D. Tuncer, A. Khanlari, F. Afshari, A. Sözen, E. Çiftçi, B. Kusun, I. Şahinkesen, Experimental and numerical analysis of a grooved hybrid photovoltaic-thermal solar drying system, Appl. Therm. Eng. 218 (2023), 119288, https://doi.org/ 10.1016/j.applthermaleng.2022.119288.
- [192] A. Hassan, A.M. Nikbahkt, Z. Welsh, P. Yarlagadda, S. Fawzia, A. Karim, Experimental and thermodynamic analysis of solar air dryer equipped with Vgroove double pass collector: techno-economic and exergetic measures, Energy Convers. Manag. 16 (2022), 100296, https://doi.org/10.1016/j. ecmx.2022.100296.
- [193] E. Abedini, H. Hajebzadeh, M.A. Mirzai, A.A. Alahdadi, H.M. Ahmadi, M. A. Salehi, M. Zakeri, Evaluation of operational parameters for drying shrimps in a cabinet hybrid dryer, Sol. Energy 233 (2022) 221–229, https://doi.org/10.1016/ j.solener.2022.01.045.
- [194] R. Rebhi, Y. Menni, G. Lorenzini, H. Ahmad, Forced-convection heat transfer in solar collectors and heat exchangers: a review, J. Adv. Res. Appl. Sci. Eng. Technol. 26 (2022) 1–15, https://doi.org/10.37934/araset.26.3.115.
- [195] A.B. Loemba, B. Kichonge, T. Kivevele, Comprehensive assessment of heat pump dryers for drying agricultural products, Energy Sci. Eng. (2022) 1–30, https://doi. org/10.1002/ese3.1326.
- [196] S. Mirzaei, M. Ameri, H. Mortezapour, Comparative energy-exergy and economicenvironmental analyses of recently advanced solar photovoltaic and photovoltaic thermal hybrid dryers: a review, Dry. Technol. (2022) 1–52, https://doi.org/ 10.1080/07373937.2022.2113793.
- [197] A.K. Bhardwaj, R. Kumar, S. Kumar, B. Goel, R. Chauhan, Energy and exergy analyses of drying medicinal herb in a novel forced convection solar dryer integrated with SHSM and PCM, Sustain. Energy Technol. Assessments 45 (2021), 101119, https://doi.org/10.1016/j.seta.2021.101119.
- [198] M.C. Gilago, V.R. Mugi, V.P. Chandramohan, Investigation of exergy-energy and environ-economic performance parameters of active indirect solar dryer for pineapple drying without and with energy storage unit, Sustain. Energy Technol. Assessments 53 (2022), 102701, https://doi.org/10.1016/j.seta.2022.102701.
- [199] S. Madhankumar, K. Viswanathan, Computational and experimental study of a novel corrugated-type absorber plate solar collector with thermal energy storage moisture removal device, Appl. Energy 324 (2022), 119746, https://doi.org/ 10.1016/j.apenergy.2022.119746.
- [200] D. Chaatouf, A.G. Ghiaus, S. Amraqui, Optimization of a solar air heater using a phase change material for drying applications, J. Energy Storage 55 (2022), 105513, https://doi.org/10.1016/j.est.2022.105513.
- [201] Mohammadhossein Rezaei, Mohammad Sefid, Khalid Almutairi, Mostafaeipour Ali, Hoa Xuan Ao, Seyyed Jalaladdin Hosseini Dehshiri, Seyyed Shahabaddin Hosseini Dehshiri, Shahariar Chowdhury, Kuaanan Techato, Investigating performance of a new design of forced convection solar dryer, Sustain. Energy Technol. Assessments 50 (2022), 101863, https://doi.org/ 10.1016/j.seta.2021.101863.
- [202] Z. Tagnamas, H. Lamsyehe, H. Moussaoui, Y. Bahammou, M. Kouhila, A. Idlimam, A. Lamharrar, Energy and exergy analyses of carob pulp drying system based on a solar collector, Renew. Energy 163 (2021) 495–503, https://doi.org/10.1016/j. renene.2020.09.011.
- [203] A. Khanlari, A. Sözen, F. Afshari, A.D. Tuncer, Energy-exergy and sustainability analysis of a PV-driven quadruple-flow solar drying system, Renew. Energy 175 (2021) 1151–1166, https://doi.org/10.1016/j.renene.2021.05.062.
- [204] A. Sharma, C.R. Chen, N.V. Lan, Solar-energy drying systems: a review, Renew. Sustain. Energy Rev. 13 (2009) 1185–1210, https://doi.org/10.1016/j. rser.2008.08.015.
- [205] K.H. Solangi, M.R. Islam, R. Saidur, N.A. Rahim, H. Fayaz, A review on global solar energy policy, Renew. Sustain. Energy Rev. 15 (2011) 2149–2163, https:// doi.org/10.1016/j.rser.2011.01.007.
- [206] J.M. Kissel, S.C. Krauter, Adaptations of renewable energy policies to unstable macroeconomic situations—case study: wind power in Brazil, Energy Pol. 34 (2006) 3591–3598, https://doi.org/10.1016/j.enpol.2005.07.013.

- [207] J. Lipp, Lessons for effective renewable electricity policy from Denmark, Germany and the United Kingdom, Energy Pol. 35 (2007) 5481–5495, https://doi.org/ 10.1016/j.enpol.2007.05.015.
- [208] F. Hvelplund, Political prices or political quantities, New energy 5 (2001) 18-23.
- [209] V. Lauber, REFIT and RPS: options for a harmonised Community framework, Energy Pol. 32 (2004) 1405–1414, https://doi.org/10.1016/S0301-4215(03) 00108-3.
- [210] P. Ekins, Step changes for decarbonising the energy system: research needs for renewables, energy efficiency and nuclear power, Energy Pol. 32 (2004) 1891–1904, https://doi.org/10.1016/j.enpol.2004.03.009.
- [211] H. Du, N. Li, M.A. Brown, Y. Peng, Y. Shuai, A bibliographic analysis of recent solar energy literatures: the expansion and evolution of a research field, Renew. Energy 66 (2014) 696–706, https://doi.org/10.1016/j.renene.2014.01.018.
- [212] B. Mcelroy Michael, X. Chen, Wind and solar power in the United States: status and prospects, CSEE J. Power Energy Sys 3 (2017) 1–6, https://doi.org/ 10.17775/CSEEJPES.2017.0002.
- [213] T. Muneer, M. Asif, S. Munawwar, Sustainable production of solar electricity with particular reference to the Indian economy, Renew. Sustain. Energy Rev. 9 (2005) 444–473, https://doi.org/10.1016/j.rser.2004.03.004.
- [214] G. Shrimali, J. Kniefel, Are government policies effective in promoting deployment of renewable electricity resources? Energy Pol. 39 (2011) 4726–4741, https://doi.org/10.1016/j.enpol.2011.06.055.
- [215] T. Berry, M. Jaccard, The renewable portfolio standard:: design considerations and an implementation survey, Energy Pol. 29 (2001) 263–277, https://doi.org/ 10.1016/S0301-4215(00)00126-9.
- [216] J.G. Pieters, J.M. Deltour, Modelling solar energy input in greenhouses, Sol. Energy 67 (1999) 119–130, https://doi.org/10.1016/S0038-092X(00)00054-
- [217] U. Büsgen, W. Dürrschmidt, The expansion of electricity generation from renewable energies in Germany: a review based on the Renewable Energy Sources Act Progress Report 2007 and the new German feed-in legislation, Energy Pol. 37 (2009) 2536–2545, https://doi.org/10.1016/j.enpol.2008.10.048.
- [218] L. Dusonchet, E. Telaretti, Economic analysis of different supporting policies for the production of electrical energy by solar photovoltaics in western European Union countries, Energy Pol. 38 (2010) 3297–3308, https://doi.org/10.1016/j. enpol.2010.01.053.
- [219] M. McHenry, Policy options when giving negative externalities market value: clean energy policymaking and restructuring the Western Australian energy sector, Energy Pol. 37 (2009) 1423–1431, https://doi.org/10.1016/j. enpol.2008.12.009.
- [220] L. Fang, X. Honghua, W. Sicheng, National survey report of PV power applications in China, Int. Energy Agency (2014) 1–45. https://iea-pvps.org/wp-content/upl oads/2020/01/nsr_2012_chn.pdf.
- [221] J. Byrne, L. Kurdgelashvili, D. Poponi, A. Barnett, The potential of solar electric power for meeting future US energy needs: a comparison of projections of solar electric energy generation and Arctic National Wildlife Refuge oil production, Energy Pol. 32 (2004) 289–297, https://doi.org/10.1016/S0301-4215(03)00107-1.
- [222] N. Martin, J. Rice, The solar photovoltaic feed-in tariff scheme in New South Wales, Australia, Energy Pol. 61 (2013) 697–706, https://doi.org/10.1016/j. enpol.2013.06.038.
- [223] L.K. Wiginton, H.T. Nguyen, J.M. Pearce, Quantifying rooftop solar photovoltaic potential for regional renewable energy policy, Comput. Environ. Urban Syst. 34 (2010) 345–357, https://doi.org/10.1016/j.compenvurbsys.2010.01.001.
- [224] N. Caldés, M. Varela, M. Santamaría, R. Sáez, Economic impact of solar thermal electricity deployment in Spain, Energy Pol. 37 (2009) 1628–1636, https://doi. org/10.1016/j.enpol.2008.12.022.
- [225] M. Zarezade, A. Mostafaeipour, Identifying the effective factors on implementing the solar dryers for Yazd province, Iran, Renew. Sustain. Energy Rev. 57 (2016) 765–775, https://doi.org/10.1016/j.rser.2015.12.060.
- [226] A. Aboltins, J. Palabinskis, New types of air heating solar collectors and their use in drying agricultural products, Agron. Res. 11 (2013) 267–274.
- [227] A.K. Srivastava, S.K. Shukla, S. Mishra, Evaluation of solar dryer/air heater performance and the accuracy of the result, Energy Proc. 57 (2014) 2360–2369, https://doi.org/10.1016/j.egypro.2014.10.244.
- [228] S.F. Dina, H. Ambarita, F.H. Napitupulu, H. Kawai, Study on effectiveness of continuous solar dryer integrated with desiccant thermal storage for drying cocoa beans, Case Stud. Therm. Eng. 5 (2015) 32–40, https://doi.org/10.1016/j. csite.2014.11.003.
- [229] P. Das, C. Velayudhan Parvathy, A critical review on solar chimney power plant technology: influence of environment and geometrical parameters, barriers for commercialization, opportunities, and carbon emission mitigation, Environ. Sci. Pollut. Res. 29 (2022) 69367–69387, https://doi.org/10.1007/s11356-022-22623-7.
- [230] S. Benmarraze, D. Itskhokine, M. Benmarraze, E. Alasis, M. Tawalbeh, W. Shahin, Status of implementation of the first Linear Fresnel solar thermal power plant in the Middle East–WECSP solar project in the Kingdom of Jordan, Energy Proc. 69 (2015) 1586–1596, https://doi.org/10.1016/j.egypro.2015.03.113.
- [231] M. Chandel, G.D. Agrawal, S. Mathur, A. Mathur, A, Techno-economic analysis of solar photovoltaic power plant for garment zone of Jaipur city, Case Stud. Therm. Eng. 2 (2014) 1–7, https://doi.org/10.1016/j.csite.2013.10.002.
- [232] O. Prakash, A. Kumar, Solar greenhouse drying: a review, Renew. Sustain. Energy Rev. 29 (2014) 905–910, https://doi.org/10.1016/j.rser.2013.08.084.

- [233] ISES, Solar World Forum, Brighton, UK, Crop Drying with Solar Air Heaters in
- Tropical Nigeria, 1981, pp. 997–1005.
 [234] Ali Mostafaeipour, Mojgan Zarezade, Hossein Rezaei, Hamid Reza Arabnia, Sedaghat Ahmad, Using factor analysis to identify effective factors to implement

solar dryers: a case study, Int. Conf. Industrial Eng. Operations Manag. Rabat, Morocco (2017) 2302–2313.

[235] Yoo Ji Ye, HeeJung Kim, EunJi Woo, ChanJin Park, On solar energy utilization for drying technology, Int. J. Environ. Sustain Dev. 8 (2017) 305–311, https://doi. org/10.18178/ijesd.2017.8.4.968.